Journal of Accounting and Public Policy xxx (xxxx) xxx-xxx

Contents lists available at ScienceDirect



Journal of Accounting and Public Policy

journal homepage: www.elsevier.com/locate/jaccpubpol



# Full length article The influence of business strategy on annual report readability

Edwin KiaYang Lim<sup>a</sup>, Keryn Chalmers<sup>b,\*</sup>, Dean Hanlon<sup>c</sup>

<sup>a</sup> Deakin Business School, Deakin University, Melbourne Burwood Campus, VIC 3125, Australia

<sup>b</sup> Swinburne Business School, Swinburne University of Technology, Mail H25, PO Box 218, Hawthorn, VIC 3122, Australia

<sup>c</sup> Monash Business School, Monash University, Sir John Monash Drive, Caulfield East, VIC 3145, Australia

#### ARTICLE INFO

Keywords: Annual report readability Business strategy Fog index

#### ABSTRACT

Concerns about the complexity of firm disclosures have prompted regulators to initiate projects to improve the readability of annual reports. We investigate business strategy as a determinant of annual report readability. As business strategy fundamentally determines a firm's product and market domain, technology, and organizational structure, it influences a firm's operating complexity, environmental uncertainty and information asymmetry. Consequently, business strategy frames the level, wording, and complexity of disclosures. We capture a firm's business strategy based on the Miles and Snow (1978) strategic typology and measure 10-K readability with Li's (2008) Fog index. We find that firms pursuing an innovation-oriented prospector strategy have less readable 10-Ks relative to firms pursuing an efficiency-oriented defender strategy. We also find that prospectors display more negative and uncertainty tones while defenders as it suggests that efforts to improve annual report readability may be limited for some firms given that business strategy is a fundamental determinant of readability and pronouncements accommodating different strategic orientations are not feasible.

#### 1. Introduction

As the scope of reporting requirements has widened to a broader narrative including corporate governance, sustainability and responsibility, remuneration, enhanced disclosure regimes and long form audit reports, the complexity and materiality of information in annual reports is being questioned. This has prompted the international financial reporting community to initiate projects on streamlining annual reports to improve their readability. Responses in the US include the Securities and Exchange Commission (SEC)'s 1998 Plain English Mandate (SEC Rule 421 (d)) and the Financial Accounting Standards Board (FASB)'s ongoing Disclosure Framework project, initiated in 2009, with the objective of making financial statement disclosures more effective, coordinated and less redundant. Disclosure reform is also on the agenda of the International Accounting Standards Board (IASB) and the UK Financial Reporting Council (FRC).<sup>1</sup> Form 10-Ks report accounting numbers prepared in accordance with accounting standards, as well as unstructured textual narratives disclosing accounting policies and describing business operations and financial performance. When preparing annual reports, including the financial reports and accompanying notes, and the management discussion and analysis (MD &A) sections, managers have discretion over the content emphasized, and the language and writing style used in their narratives (Henry and Leone, 2016; Loughran and McDonald, 2014). Li (2010) argues that unstructured textual narratives in annual reports

\* Corresponding author.

E-mail addresses: e.lim@deakin.edu.au (E.K. Lim), kchalmers@swin.edu.au (K. Chalmers), dean.hanlon@monash.edu (D. Hanlon).

<sup>1</sup> Better Communication in Financial Reporting is a key pillar of the IASB's work agenda for 2017-2021. In March 2017, the IASB issued Discussion Paper DP/2017/1 Disclosure Initiative – Principles of Disclosure.

https://doi.org/10.1016/j.jaccpubpol.2018.01.003

0278-4254/ © 2018 Elsevier Inc. All rights reserved.

#### E.K. Lim et al.

#### Journal of Accounting and Public Policy xxx (xxxx) xxx-xxx

exhibit irregularities, ambiguities, and managerial opportunism. Form 10-K volume of pages and footnote disclosures has also increased, up 16 percent and 28 percent respectively over the six-year period 2004–2010, with evidence of disclosure repetition and redundancy (Iannoconi and Sinnett, 2011). Investors rely on information in 10-Ks to infer trading and pricing decisions (Cazier and Pfeiffer, 2016; Griffin, 2003; Huddart et al., 2007) and complex disclosures, including less readable annual reports, can impair users' ability to process information, and hence affect their judgment and decision making (Li, 2008; Miller, 2010; You and Zhang, 2009). Consequently, policy makers and various capital market participants are increasingly concerned about the readability of 10-Ks that underpins the understandability and decision usefulness of information contained therein.

Such concern has spawned research on the association between readability and individual firm characteristics synonymous with operating complexity and uncertainty.<sup>2</sup> In this study, we investigate whether business strategy explains 10-K readability. By fundamentally influencing a firm's operating complexity and environmental uncertainty (Bentley et al., 2013; Hambrick, 1983a; Higgins et al., 2015; Miles and Snow, 1978), business strategy warrants attention in explaining 10-K readability as it is a comprehensive measure that captures firm-level complexity not necessarily captured by individual firm characteristics. As Bentley et al. (2013) suggest, business strategy is a broad underlying measure that "captures a construct that is greater than the sum of its parts" (p.805).

Specifically, we explore the 10-K readability of innovation-oriented prospector firms and efficiency-oriented defender firms. Pursuing innovation increases firms' exposure to operating complexity and environmental uncertainty due to the risky nature of research and development (R&D), and unpredictable and constantly changing consumer choices (Miles and Snow, 1978; Naiker et al., 2008), which can increase disclosure complexity and incentives to obfuscate information. Conversely, firms pursuing a more stable cost-efficient strategy relying less on innovation, are less exposed to operating complexity, environmental uncertainty, and costly failure, resulting in less complex disclosures and lower incentives to obfuscate information. Our proposition is that prospectors have less readable 10-Ks relative to defenders.

With the accounting standard setting community focused on improving the readability of financial reports, our study is motivated by considering the influence of strategy on communication and how this may restrict the success of reporting reforms to a subset of firms. White (2013), former SEC Chair, notes the importance of broadening considerations of disclosure complexity as accounting standards and volume of mandated disclosures may not be the sole contributing factor. Zahra et al. (2005) and Bentley et al. (2013) highlight the importance of establishing direct antecedents of financial reporting outcomes rather than merely identifying their perfunctory indicators. Bentley et al. (2013) associate prospectors with financial reporting irregularities, and Bentley Goode et al. (2018) relate prospectors to a richer external information environment. Our study complements these studies by investigating whether business strategy influences the readability and tone of 10-K information.

We classify a firm's strategic orientation using Miles and Snow (1978, 2003) typology and following the methodology of Bentley et al. (2013) which relies on a collective of firm characteristics: research intensity, marketing and advertising efforts, historical growth, operational efficiency, capital intensity, and organizational stability. We obtain the Fog index, a proxy for 10-K readability, from Feng Li's website.<sup>3</sup> Using 24,817 firm-year observations spanning 1994–2011, we find that prospectors have less readable 10-Ks relative to defenders. Investigating if business strategy influences disclosure sentiment or tone using Loughran-McDonald sentiment measures, we find that prospectors display greater negative and uncertainty tones while defenders use more litigious tone in their 10-Ks. Greater use of negative and uncertainty tones in the 10-Ks can explain the lower readability of prospectors' 10-Ks. Our results are robust to using alternative measures of readability including the Kincaid and Flesch indexes. We do not, however, find an association between business strategy and 10-K length, suggesting that linguistic index and length capture different facets of reporting and disclosure complexity. Further, we show that in both sub-samples of good and poor performing firms, prospectors have less readable 10-Ks relative to defenders, suggesting that business strategy plays a more fundamental role than earnings performance in explaining 10-K readability.

Our study offers insights for policy makers seeking to enhance the readability and reduce the complexity of annual reports. Christopher Cox (2007), former SEC Chair, suggests the use of the Fog index to measure compliance with the SEC's plain English initiatives.<sup>4</sup> Our study suggests that policy makers' efforts to improve and assess annual report readability may be limited to influencing a subset of firms with a strategic orientation not requiring, or incentivizing, complex communications.

Our study also contributes to the readability and business strategy literatures. We document that business strategy is associated with annual report readability, controlling for individual firm complexity variables. Cazier and Pfeiffer (2016) show a firm's operating complexity, disclosure redundancy, and residual disclosure (i.e. 10-K length unexplained by a firm's operating complexity or redundancy) as determinants of 10-K length. However, amongst the firm characteristics examined, Cazier and Pfeiffer (2016) find that R&D does not explain variation in 10-K length. We show that while business strategy (including R&D) is not associated with 10-K length, it is associated with 10-K readability. Our study also provides a plausible explanation to Bentley Goode et al.'s (2018) finding that prospectors (defenders) have higher (lower) analyst following since lower (greater) 10-K readability can increase (reduce) the demand for analyst services. Our findings also imply that strategy should be controlled for in future studies to mitigate the concern of omitted correlated variables (Li, 2010, p. 158).

The remainder of this paper proceeds as follows. Section 2 reviews the literature on annual report readability and business strategy, and formulates the hypothesis to be tested. Section 3 presents the research design. Section 4 reports sample selection,

<sup>&</sup>lt;sup>2</sup> Such individual firm characteristics include firm size, market to book ratio, R&D, age, and geographic and operating segments (Cazier and Pfeiffer, 2016; Kumar, 2014; Laksamana et al., 2012; Li, 2008; Lo et al. 2017).

<sup>&</sup>lt;sup>3</sup> The data for 10-K readability are made available by Feng Li at http://webuser.bus.umich.edu/feng/.

<sup>&</sup>lt;sup>4</sup> The UK FRC also encourages the use of plain English to enhance financial information understandability (Financial Reporting Council, 2011).

#### E.K. Lim et al.

#### Journal of Accounting and Public Policy xxx (xxxx) xxx-xxx

descriptive statistics and the main results. Sections 5 and 6 present robustness and additional analyses, respectively. Section 7 concludes the study.

#### 2. Literature review and hypothesis development

In the fields of communication, psychology, and education, readability is defined as the ease of understanding a message attributable to the style of writing (Barnett and Leoffler, 1979). Corporate disclosures, including those within the annual report, contain significant unstructured textual information (Li, 2010). Prior studies suggest that investors find the annual report difficult to read (Bartlett and Chandler, 1997; Courtis, 1982; Hynes and Bexley, 2004; Lee and Tweedie, 1975). Davis and Tama-Sweet (2012) claim that managers have discretion over the degree of detail (or content), as well as the words used to describe a particular data item or event, in narrative disclosures. Beattie (2014) provides an extensive review on accounting narratives. Although narrative disclosures can improve transparency and meet stakeholders' information needs, there are concerns that narrative disclosures may lose utility or obfuscate issues with writing styles that confuse, distract or perplex readers (Courtis, 2004; Rutherford, 2003; SEC, 1998).

According to comprehension theory, readability affects individuals' understanding of the disclosures and, subsequently, their judgments (Kintsch and Van Dijk, 1978; Masson and Waldron, 1994). Rennekamp (2012) and Shah and Oppenheimer (2007) claim that processing fluency, reflective of individuals' concern about the ease of processing information, determines their reliance on the disclosure. As per the incomplete revelation hypothesis, investors react less timely and completely to information that is more difficult to process (Bloomfield, 2002; Grossman and Stiglitz, 1980). Complex information requires investors to exert more cognitive efforts. This reduces their understanding and ability to evaluate the firm's prospects based on the information, and can impair their decision making (Lee, 2012).

The Fog (Gunning-Fog) index is a proxy for annual report readability used by prior studies (Biddle et al., 2009; Lawrence, 2013; Lehavy et al., 2011; Li, 2008; Miller, 2010). The Fog index is inversely associated with annual report readability. Extant studies have documented the capital market effects of annual report readability suggesting that less readable reports (including 10-Ks): discourage and impede informed securities trading by small investors (Hirshleifer and Teoh, 2003; Lawrence, 2013; Miller, 2010); exacerbate market under-reaction (You and Zhang, 2009); prolong the price discovery process and lengthen post-filing drifts (Lee, 2012); indirectly reduce investors' perceptions of management credibility (Rennekamp, 2012); and induce more analyst following but result in greater analyst forecast dispersion and error (Lehavy et al., 2011; Loughran and McDonald, 2011).

Extant studies have also explored the determinants of 10-K readability. Li (2008) associates lower earnings with less readable 10-Ks, with attribution to the management obfuscation hypothesis which posits that managers may use vague and complex language to downplay adverse information and avoid negative market reactions (Courtis, 1998; Dempsey et al., 2012). Bloomfield (2008) notes alternative explanations to Li's (2008) findings. For instance, according to the attribution theory in psychology, investors expect firms to explain bad news and this is more complicated to convey. Kumar (2014) investigates the effect of culture on annual report readability. He finds that US-listed Asian firms whose domestic culture is more secretive (i.e. preference for confidentiality and the restriction of disclosure) exhibit less readable annual reports, whereas those with higher ownership dispersion provide more readable annual reports to mitigate increased conflicts of interest between owners under such an ownership structure. Laksamana et al. (2012) show that over-compensated top management has less readable Compensation Discussion and Analysis (CD&A) due to obscuring compensation practices that are disconnected with economic determinants of pay. Lo et al. (2017) show that firms that have managed earnings to outperform the previous year's earnings tend to obfuscate bad performance and earning management, rendering the MD& A section of 10-Ks more difficult to read. Cazier and Pfeiffer (2016) find 10-K length is associated with some individual firm characteristics synonymous with operating complexities (e.g., acquisition incidences, capital leases, return volatility, number of operating segments, leverage, size, free cash flow and special items) but not others (e.g., R&D, intangibles, operating leases, number of geographic segments).

We expand the literature exploring the determinants of annual report readability by proposing that business strategy is an underlying factor of 10-K readability capturing complexity and uncertainty not captured by individual firm characteristics. Business strategy, determined at a very early stage of a firm's life cycle, requires long-term resource commitments, and thus it directs how firms achieve, maintain or enhance their performance (Bentley et al., 2013; Burgelman, 2002; Snow and Hambrick, 1980; Zajac and Shortell, 1989). To maintain competitiveness, firms align their respective strategies with different patterns of product and market domain, technology, and organizational structure and process (Miles and Snow, 1978, 1984, 1986, 2003; Miles et al., 1978; Milgrom and Roberts, 1995; Navissi et al., 2017). As such, business strategy fundamentally influences a firm's operating complexity and environmental uncertainty (Bentley et al., 2013; Hambrick, 1983; Higgins et al., 2015; Miles and Snow, 1978), which can frame the level, wording, and complexity of disclosures to stakeholders in documents such as the 10-K.

We compare 10-K readability for the contrasting innovation-oriented prospector and efficiency-oriented defender strategies. Prospectors aggressively change their technologies to explore and launch new products and market opportunities. Prior literature suggests that market uncertainty, technological uncertainty, and intensity of competition determine environmental uncertainty (Jaworski and Kohli, 1993; Kohli and Jaworski, 1990). Environmental uncertainty refers to the degree of change that influences a firm's organizational design and operations, which is driven by unpredictable and constantly changing actions of customers, suppliers, competitors and regulatory groups (Child, 1972; Dess and Beard, 1984; Ghosh and Olsen, 2009). Unique, unstructured, and experimental R&D activities lead to rapid technological changes (Clark, 1985; Ecker et al., 2013; Hill et al., 2000; Kohli and Jaworski, 1990; Saviotti, 1998). Further, fierce competition and unpredictable consumer choices make it difficult to predict the outcome for highly idiosyncratic products in volatile, constantly changing environments (Song and Montoya-Weiss, 2001; Su et al., 2010). As such, prospectors are exposed to complexity and environmental uncertainty (DeSarbo et al., 2005; Ittner et al., 1997; Ittner and

#### E.K. Lim et al.

Larcker, 2001; Kotha and Nair, 1995; Williamson, 1979). Li (2008) argues that operating complexity and environmental uncertainty reduce 10-K readability.

Since environmental uncertainty induces firm performance variability (Cheng and Kesner, 1997), it accentuates information asymmetry between the firm and external stakeholders (Ghosh and Olsen, 2009), reduces the clarity of information disclosed by managers (Leifer and Mills, 1996), and complicates managerial assumptions in financial reporting (Barth et al., 2001; Bentley et al., 2013). This triggers more complex disclosures for prospectors. For instance, Dedman et al. (2009) posit that innovative firms have incentives to provide more disclosures on their R&D but acknowledge that these disclosures contain scientific complexity and technical words that make the communication of these activities difficult. In addition, since innovative activities are prone to costly failure (Huang et al., 2008; Saleh and Wang, 1993), according to attribution theory in psychology (Bloomfield, 2008) prospectors are expected to accompany any bad news associated with failures with more exhausting explanations.<sup>5</sup> Thus, *a priori*, prospectors' disclosures are more complex, technical, and contain more uncertain words that reduce 10-K readability.

Bentley (2012) and Bentley et al. (2013) suggest that prospectors exhibit greater incentives (e.g., rapid and sporadic growth, greater need for external financing, and poorer financial performance), opportunities (e.g., lack of monitoring, internal control deficiencies, less stable and more complex organizational structure), and rationalizations (e.g., individualistic and egoism based ethical climate and culture) to misreport, relative to defenders. Recent studies associate prospectors with greater financial reporting irregularities and higher audit fees (Bentley et al., 2013), aggressive tax avoidance (Higgins et al., 2015), weaker internal control over financial reporting (ICFR) (Bentley-Goode et al., 2017) and, consequently, greater likelihood of receiving a material weakness opinion (Chen et al., 2017), and value-decreasing overinvestment (Navissi et al., 2017). Lo et al. (2017) associate obfuscation of earnings management with less readable annual reports. Thus, prospectors are also likely to obfuscate unfavourable information with vague, perplexing, and complex words or sentence structure in annual reports and this impression management reduces 10-K readability.

In contrast, defenders seek to offer cost-efficient and high-quality products to customers in market niches, with limited focus on R &D, thereby reducing operating complexity and environmental uncertainty (Miles and Snow, 1978, 2003; Naiker et al., 2008). To achieve this, they continuously enhance the efficiency of technologies and facilitate stringent monitoring and control with a centralized organizational structure (Ittner and Larcker, 1997; Thomas and Ramaswamy, 1996). Consequently, defenders grant less discretion to managers and employees, whom are subject to task programmability and strict monitoring and control, which can curb opportunism and information asymmetry (Bentley et al., 2013; Ittner et al., 1997; Naiker et al., 2008).

Some studies suggest that managers engage in strategic reporting to protect against proprietary costs from competition (Bagnoli and Watts, 2010; Ellis et al., 2012). Competitors can deduce the cost function, and hence the organization and production process, of a leader from financial information such as performance and cost measures (Sadka, 2004), allowing them to subsequently formulate their competitive strategies (Beatty et al., 2013). To withstand or discourage competition for their price sensitive products (Hambrick, 1983b; Tansey et al., 2014), defenders may be incentivized to engage in strategic reporting by omitting disclosures (i.e. less voluntary disclosures), or comply with the mandatory disclosure requirements, but with an attempt to diffuse information (Wagenhofer, 1990). For example, defenders can make the disclosures vague or proactively discuss bad news and downplay good news in their 10-Ks. Wagenhofer (1990) proposes that disclosures of unfavourable information can deter opponents, and Dhaliwal et al. (2014) document that managers battle the high threat of new entrants by reporting losses more quickly than gains. This, in turn, can add noise to defenders' 10-Ks and render their 10-Ks less readable. Alternatively, due to their aversion to risk (Navissi et al., 2017), defenders may disclose surplus information to ensure compliance with mandatory reporting requirements, and disclose more bad news over good news to manage financial statement users' perceptions, with such profuse disclosures adding noise to defenders' 10-Ks. In relation to proprietary costs, Seavey et al. (2017), however, suggest that such costs are higher for prospectors. This drives strategic reporting with the consequence being the production of less comparable reports. On balance, we expect defenders to have less complex disclosures and lower incentives to obfuscate information, with a corresponding more readable 10-K, than prospectors. Formally stated, the hypothesis to be tested is:

Prospectors have less readable 10-Ks relative to defenders.

#### 3. Research design

We employ regression Model 1 to test the association between business strategy and annual report readability.

$$FOG_{it} = \beta_0 + \beta_1 BOS_{it} + \beta_2 LAGFOG_{it} + \beta_3 ABDA_{it} + \beta_4 REAM_{it} + \beta_5 EARN_{it} + \beta_6 SIZE_{it} + \beta_7 MTB_{it} + \beta_8 AGE_{it} + \beta_9 SIAT_{it} + \beta_{10}RETVOL_{it} + \beta_{11} EVOL_{it} + \beta_{12} NBSEG_{it} + \beta_{13} NGSEG_{it} + \beta_{14} NITEMS_{it} + \beta_{15} SEO_{it} + \beta_{16} MA_{it} + \beta_{17} DLW_{it} + INDUSTRY FIXED EFFECTS + YEAR FIXED EFFECTS + \varepsilon_{it}$$
(Model 1)

#### 3.1. Dependent variable

FOG, our dependent variable, is a proxy for 10-K readability obtained from Feng Li's website. Li (2008) estimates FOG as (wordsper-sentence + percent-of-complex-words) \* 0.4 where words with three syllables or more are deemed as complex words. A higher

<sup>&</sup>lt;sup>5</sup> For instance, managers may attribute unsuccessful investments to bad luck, such as external events, which entail additional length and complicated sentences to tie poor performance to those events (Miller and Ross, 1975).

#### E.K. Lim et al.

(lower) value of *FOG* suggests that the annual report is less (more) readable since it requires more (less) years of formal education for a reader of average intelligence to read the text once and understand the context.<sup>6</sup>

#### 3.2. Test variable

We follow Bentley et al. (2013) and Higgins et al. (2015) to operationalize business strategy (*BOS*), our test variable, based on the Miles and Snow (1978) strategic typology. *BOS* is a discrete score ranging from 6 to 30, constructed as the sum of quintile ranks of six variables per SIC 2-digit industry and year. The six variables are research intensity, marketing and advertising efforts, historical growth, operational efficiency, capital intensity, and organizational stability. Following Bentley et al. (2013) and Ittner et al. (1997), all variables are computed using a rolling prior five-year period over 1989–2011 for 1994–2011 business strategy scores. Existing literature shows that firms commit their resources and attempt to maintain their strategic positions over a long-term period to gain competitive advantage (Miles and Snow, 1978, 2003). Therefore, using a longer rolling period to compute the variables can minimize the random influence of external events on each variable computed to more accurately capture business strategy (Thomas and Ramaswamy, 1996; Zajac and Shortell, 1989).

Research intensity (*RDS5*), the ratio of R&D expenditures to sales, captures a firm's tendency to search for new products and markets (Naiker et al., 2008). Prospectors increase R&D intensity to maintain their reputations as innovative leaders in the market (Hambrick et al., 1983; Ittner and Larcker, 1997). In contrast, defenders have limited focus on R&D since they emphasize continuous improvement of existing products with standardized inputs (Smith et al., 1989). Therefore, prospectors (defenders) display a higher (lower) value in *RDS5*.

Marketing and advertising efforts (*SGA5*), the ratio of selling, general and administrative expenses (SG&A) to sales, reflect a firm's focus on exploiting and marketing new products (Bentley et al., 2013).<sup>7</sup> To introduce customers to new differentiated products, prospectors heavily rely on marketing and advertising (Ittner et al., 1997; Levy, 1989). In contrast, regular customers are already familiar about defenders' well-established cost-efficient products, therefore defenders rely less on marketing and advertising (Kotha and Nair, 1995; McDaniel and Kolari, 1987). Prospectors (defenders) exhibit a higher (lower) value of *SGA5*.

The third variable included is historical growth or investment opportunities (*REV5*), measured as one-year percentage change in total sales (Bentley et al., 2013). Prospectors tend to experience high growth rates by market expansion via regularly pioneering new products. In contrast, defenders stick to the existing stable market domain and achieve growth through market penetration (Said et al., 2003; Shortell and Zajac, 1990). Smith et al. (1989) juxtapose that the growth pattern of defenders is stable, while that of prospectors is spurt. Consequently, prospectors (defenders) are more likely to have a higher (lower) value of *REV5*.

Operational efficiency (*EMPS5*), the number of employees to sales, indicates a firm's ability to produce products and services efficiently (Higgins et al., 2015). Highly standardized business operations and clearly stipulated procedures enable employees in defenders to generate a higher level of sales, relative to the same number of employees in prospectors who engage in innovative activities characterized with uncertainty and loosely defined procedures that can lead to wasteful activities (Ittner et al., 1997; Naiker et al., 2008). Thus, it is expected that prospectors (defenders) have a higher (lower) *EMPS5* ratio.

The fifth variable is capital intensity (*CAP5*), reflecting a firm's technological efficiency and measured as net property, plant, and equipment to total assets (Bentley et al., 2013). Defenders are more automated and capital intensive to achieve inputs minimization and outputs maximization, that ultimately leads to economies of scale (Langerak et al., 1999; McDaniel and Kolari, 1987). In contrast, prospectors, which continuously explore new market opportunities, are less automated and capital-intensive, pay less attention to technology improvement, and are more flexible and regularly replace their technologies (Hambrick, 1983a; Segev, 1989). Therefore, a higher (lower) *CAP5* for defenders (prospectors) is expected.

Finally, organizational stability (*TEMP5*) is measured as the standard deviation of the total number of employees (Bentley et al., 2013). Employees in prospectors have shorter tenure as they move across firms based on the availability of projects, and also because they possess general skills that afford them mobility between firms. Further, senior management in prospectors can be hired externally (Thomas and Ramaswamy, 1996). Conversely, employees in defenders do not generally possess a wide range of skills allowing them to move across firms and they receive familiarisation training on the firm's business operations, rendering them more 'sticky' to the firm (Naiker et al., 2008). Further, senior management is usually promoted within defenders as intimate knowledge of the firm and its production capacity is necessary (Navissi et al., 2017; Thomas and Ramaswamy, 1996).

The six strategy variables are quintile ranked per year and two-digit SIC industry to acknowledge the co-existence of different strategies in the same industry, and to control for potential year-industry effects so that less biased quintiles can be constructed. *RDS5*, *SGA5*, *REV5*, *EMPS5*, and *TEMP5* are ranked in an ascending manner, while *CAP5* is ranked in a descending manner, with the intention that a higher (lower) value of these variables reflects prospector (defender) characteristics. The quintile-ranked scores of

<sup>&</sup>lt;sup>6</sup> Loughran and McDonald (2014) suggest file size as a superior 10-K readability proxy than the Fog index. They claim that the Fog index is poorly specified in financial applications because it deems many multisyllabic yet well understood business texts (e.g., company) complex. Whilst Bonsall et al. (2017) acknowledge the shortcomings of the Fog index, they also question the validity of file size as a proxy for readability. Specifically, they show that file size is subject to measurement error because it does not rely on actual language used in the document, and time-series and cross-sectional variations in file size are mainly driven by the inclusion of content unrelated to the underlying text in the 10-K (e.g., HTML, XML, pdf and jpeg file attachments). With the limitations of both file size and the Fog index, Bonsall et al. (2017) propose an alternative measure based on plain English attributes. We employ three alternative proxies for 10-K readability in our robustness analysis, namely the Kincaid index, the Flesch index, and annual report length. We source these alternative measures from Li's website.

<sup>&</sup>lt;sup>7</sup> Pantzalis and Park (2009) note that less than 20 percent of firms in Compustat report their advertising expenditures. Thus, we utilize SG&A to capture firm-level marketing and advertising activities, which is in line with Bentley et al. (2013). Our results (untabulated) remain qualitatively and quantitatively similar with the use of advertising expenditure, recognizing that this specification results in a much smaller sample.

#### E.K. Lim et al.

#### Journal of Accounting and Public Policy xxx (xxxx) xxx-xxx

these six variables are summed every year, ranging from 6 to 30, to indicate a firm's business strategy: defender (*DEF*: 6–12); analyzer (*AN*: 13–23); and prospector (*PRO*: 24–30). We expect a positive coefficient on *BOS* to show that prospector-oriented firms have less readable 10-Ks (i.e. higher *FOG*). We also replace *BOS* with separate dichotomous variables for defender (*DEF*) and prospector (*PRO*) firms, and expect a negative (positive) coefficient on *DEF* (*PRO*), suggesting that defenders (prospectors) display more (less) readable 10-Ks.

#### 3.3. Control variables

Following previous readability studies (e.g. Cazier and Pfeiffer, 2016; Laksamana et al., 2012; Li, 2008; Lo et al., 2017), we control for the following determinants of annual report readability. *LAG\_FOG*, measured as one-year lagged *FOG*, controls for persistence in readability (Lo et al., 2017). Controlling for the lagged dependent variable can also attenuate the concern of reverse causality (Chen et al., 2011; Klein, 1998). We expect *LAG\_FOG* to be positively associated with *FOG*. Lo et al. (2017) associate earnings management with 10-K readability. We control for both accruals and real earnings management since Cohen et al. (2008) document increasing (decreasing) levels of real (accruals) earnings management after the passage of the Sarbanes-Oxley Act in 2002. Accruals earnings management is proxied by the absolute value of Modified Jones (1991) discretionary accruals (*ABDA*), as per Dechow et al. (1995) and Cohen et al. (2008). Real earnings management is proxied by *REAM* which is the sum of the standardized three real earnings management measures, namely abnormal cash flows from operations (*RCFO*), abnormal production costs (*RPROD*), and abnormal discretionary expense (*RDISX*), as per Cohen et al. (2008), Park (2017), and Roychowdhury (2006). We expect these variables to be positively associated with *FOG*.

*EARN* is calculated as operating earnings scaled by book value of assets. We expect a negative coefficient on *EARN* since firms with poorer performance have more perplexing disclosures to explain the bad news and/or greater incentives to obfuscate information (Bloomfield, 2008), resulting in less readable 10-Ks (i.e. higher *FOG*). *SIZE*, measured as the logarithm of the market value of equity at the end of the fiscal year, reflects the complexity of a firm's operating and business environment. *MTB*, measured as the market value of equity plus the book value of liabilities divided by the book value of assets at the end of the fiscal year, captures a firm's growth potential and, hence, complex and uncertain business models.<sup>8</sup> Positive coefficients are expected on *SIZE* and *MTB* since larger firms and high growth firms are expected to produce less readable 10-Ks. *AGE* is calculated as the logarithm of the number of years since a firm first appears in CRSP monthly stock return files. We predict a negative coefficient on *AGE* because older firms characterized with less information uncertainty and greater investor familiarity with their business models produce less complex and, hence, more readable 10-Ks. *SIAT*, measured as special items scaled by book value of assets, is expected to have a negative coefficient since firms with more negative special items exhibit more complex 10-Ks.

Business and operating environment volatility is captured with *RETVOL*, the standard deviation of monthly stock returns in the last year, and *EVOL*, the standard deviation of operating earnings in the last five fiscal years. Both measures are expected to yield positive coefficients, as more volatile environments lead to less readable 10-Ks. We control for, and expect positive coefficients on, *NBSEG* (i.e. the logarithm of the number of business segments from the Compustat segment file plus 1) and *NGSEG* (i.e. the logarithm of the number of business readable 10-Ks. We further include financial complexity proxied by *NITEMS*, measured as the logarithm of the number of non-missing items on Compustat. We predict a positive coefficient on *NITEMS* since financially complex firms disclose more and have less readable 10-Ks. Firms with seasoned equity offerings and merger and acquisitions tend to have more detailed disclosures that can result in less readable 10-Ks, thus we include and expect positive coefficients on *SEO*, a dichotomous variable that equals 1 if a firm has a merger and acquisition during the year and 0 otherwise. Firms that are incorporated in Delaware are subject to different corporate laws and, thus, exhibit different levels of readability (Daines, 2001). This is controlled for with a dichotomous variable (*DLW*) being 1 if a company is incorporated in Delaware and 0 otherwise. Finally, we control for industry and year fixed effects, and cluster standard errors at the firm level (Cameron and Miller, 2015; Petersen, 2009).

#### 4. Sample selection, descriptive statistics, and main results

Table 1 presents our sample selection procedure. We begin with 219,904 firm-years with Compustat data between 1989 and 2011 after removing firm-years with zero or negative sales and assets, and missing historical SIC. Firm-year removals are: 50,462 firm-years operating in the Utilities and Financial industries (SIC codes 49, 60–69) since they are highly regulated and subject to different accounting rules; and 107,346 firm-years with insufficient five-year rolling data to compute all six business strategy component variables. This yields 62,096 firm-years with available business strategy scores between 1994 and 2011.<sup>9</sup> A merge with 10-K *FOG* data obtained from Feng Li's website results in 38,514 firm-years. Finally, after merging with other determinants of 10-K readability estimated using Compustat and CRSP data, 24,817 firm-years remain in our sample. The sample comprises 1941 defender firm-years (eight percent), 1216 prospector firm-years (five percent), and 21,660 analyzer firm-years (87 percent). The domination of analyzer firm-year observations is consistent with prior studies, including Bentley et al. (2013), Higgins et al. (2015), and Navissi et al. (2017),

<sup>&</sup>lt;sup>8</sup> Concerned that *MTB* is highly correlated with *REV5*, as they both capture growth potential, *MTB* is excluded from the regression in a robustness test (untabulated) with quantitatively and qualitatively similar results found.

<sup>&</sup>lt;sup>9</sup> Our sample ends in 2011 because Feng Li provides FOG data up to 2011.

#### E.K. Lim et al.

Table 1
Sample selection.

Firms-years with Compustat data between 1989 and 2011 (after eliminating firm-years with zero or negative sales and assets, and missing historical SIC)	219,904
Less firm-years operating in the Utilities and Financial industries (SIC codes 49, 60-69)	(50,462)
Less firm-years with insufficient five year rolling data to compute all six business strategy component variables	(107,346)
Firm-years with available business strategy scores between 1994 and 2011	62,096
Firm-years after merging with available FOG data from Feng Li's website	38,514
Firm-years after merging with control variables for 10-K readability	24,817
Comprising	
Defender firm years	1941
Prospector firm years	1216
Analyzer firm years	21,660

as many firms pursue a hybrid business strategy.

The industry distribution (untabulated) is similar to Bentley et al. (2013) and supports Miles and Snow (1978, 2003) that three types of viable business strategies co-exist in different industries. The manufacturing sector is the largest industry segment represented in the full sample (60 percent), as well as in the sub-samples of prospectors (64 percent), defenders (74 percent) and analyzers (59 percent).

The descriptive statistics for the main variables are reported in Table 2 Panel A. Similar to Bentley et al. (2013), *BOS* has a mean (median) of 17.86 (18.00) with a standard deviation of 3.59. The statistics for *FOG* and other control variables resemble those reported by Li (2008) and Lo et al. (2017). For instance, the mean (median) *FOG* is 19.46 (19.27). The statistics for *ABDA* and *REAM* (including its three main components, namely *RCFO*, *RPROD*, and *RDISX*) are similar to those reported by Cohen et al. (2008).

We perform univariate analysis to provide preliminary insights on the association between *FOG* and *BOS*. As reported in Table 2 Panel B, the mean of *FOG* increases monotonically across *BOS* quartiles, suggesting that firms more closely resembling a prospector strategy exhibit less readable 10-Ks. The difference in mean *FOG* between the highest and lowest *BOS* quartiles is 0.166, significant at the one percent level (*t*-statistic = 5.47). Similarly, Table 2 Panel C shows that mean *FOG* increases progressively in the defender, analyzer and prospector sub-samples when firms are categorized based on their strategic type. The difference in mean *FOG* between prospector and defender sub-samples is 0.437, and is significant at the one percent level (*t*-statistic = 7.02).

We present the Pearson (upper diagonal) and Spearman (lower diagonal) correlations in Table 3. FOG and BOS are positively correlated. Further, BOS is correlated with earnings management proxies (*ABDA* and *REAM*) and many of the complexity variables previously included in readability studies. The highest variance inflation factor of 2.44, pertinent to *SIZE*, is less than the threshold of 10 thereby mitigating multicollinearity concerns among independent variables (Kennedy, 1992).

Table 4 reports the main regression results. In column (1), the coefficient on *BOS* is positive (0.016) and significant at the one percent level (*t*-statistic = 3.64). This suggests that firms with a higher business strategy score, resembling a prospector strategy, have less readable 10-Ks. In column (2), *BOS* is replaced with separate dichotomous variables for defender (*DEF*) and prospector (*PRO*) firms. We find significantly negative (positive) coefficients on *DEF* (*PRO*), suggesting that defenders (prospectors) are associated with more (less) readable 10-Ks. An *F*-test of the equality of the *DEF* and *PRO* coefficients rejects the null hypothesis that these coefficients are equal (Higgins et al., 2015). Finally, column (3) reports the results after restricting the sample to only prospector and defender firm-year observations. The coefficient on *PRO* is positive (0.324) and significant (*t*-statistic = 3.35) at the one percent level, consistent with our hypothesis that prospectors are associated with less readable 10-Ks relative to defenders. Among the control variables, *LAG\_FOG* is significantly and positively associated with *FOG* as per Lo et al. (2017), while *EARN* and *NGSEG* are significantly and negatively associated with *FOG* on a consistent basis. This suggests that business strategy as an *ex ante* determinant of a firm's operating environment, complexity, and financial reporting outcome is value adding in understanding 10-K readability over and above standalone variables used to compute business strategy (e.g., growth opportunities) and *ex post* financial reporting outcome variables such as earnings management.

#### 5. Additional analysis

#### 5.1. Disclosure sentiment

Extant studies have examined the impact of disclosure tone on firm outcomes. For example, Kothari et al. (2009) find that negativity in management's disclosures is related to greater stock return volatility of a firm. Loughran and McDonald (2013) show that initial public offerings with more uncertain text demonstrate greater first-day returns, absolute offer price revisions, and subsequent volatility. Ertugrul et al. (2017) find that firms with less readable 10-Ks and more ambiguous or uncertain tones experience higher costs of external financing and display more managerial information hoarding that results in greater stock price crash risk. However, these studies rarely examine the determinants of disclosure tones. We examine whether business strategy influences the lexical properties or disclosure tone of 10-Ks using three Loughran-McDonald sentiment measures obtained from WRDS SEC Filings Queries (Loughran and McDonald, 2011). First, litigious tone (*LIT*) refers to the Loughran-McDonald litigious word proportion within

#### Journal of Accounting and Public Policy xxx (xxxx) xxx-xxx

#### Journal of Accounting and Public Policy xxx (xxxx) xxx-xxx

#### E.K. Lim et al.

#### Table 2

Descriptive statistics and univariate analysis.

Variable		Mean	Q1	Median	Q3	Std dev.			
Panel A: Descriptive st	atistics for depender	nt, test and control varia	bles						
FOG	, ,	19.458	18.451	19.273	20.242	1.692			
BOS		17.858	15.000	18.000	20.000	3.589			
ABDA		0.069	0.019	0.044	0.085	0.087			
REAM		0.063	-0.027	0.037	0.135	0.192			
RCFO		0.016	-0.035	0.023	0.078	0.124			
RPROD		-0.038	-0.148	-0.039	0.067	0.220			
RDISX		0.085	-0.058	0.040	0.198	0.286			
EARN		0.055	0.025	0.079	0.129	0.177			
SIZE		5.886	4.361	5.898	7.273	2.119			
MTB		1.865	1.091	1.446	2.094	1.516			
AGE (raw)		18.716	9.000	15.000	28.000	12.062			
SI (raw)		- 40.688	-6.800	0.000	0.000	489.276			
RETVOL		0.143	0.085	0.121	0.175	0.092			
EVOL		0.048	0.015	0.028	0.054	0.067			
NBSEG (raw)		5.368	2.000	3.000	9.000	4.589			
NGSEG (raw)		6.503	3.000	4.000		5.219			
					9.000				
NITEMS (raw)		270.576	235.000	267.000	306.000	40.045			
SEO		0.261				0.439			
MA		0.406				0.491			
DLW		0.584				0.493			
	BOS quar	BOS quartiles							
	1	2	3	4	4-1Vv	t-statistic			
Panel B: Mean of FOG	within each busine	ss strategy score quartile	1						
Mean of FOG	19.372	19.433	19.507	19.538	0.166	5.47***			
N	6245	7484	5129	5959					
		Business strategy							
		Defender	Analyzer	Prospector	Prospector-defender	t-statisti			
Panel C: Mean of FOG	within each busine	ss strategy type							
Mean of FOG		19.279	19.460	19.716	0.437	7.02***			
N		1941	21,660	1216					

FOG represents the Fog index of the entire annual report developed by Feng Li. A higher FOG indicates that the annual report is less readable. BOS is Bentley et al. (2013) business strategy score ranging from 6 to 30, constructed as the sum of quintile ranks of the following six variables per SIC 2-digit industry and year, where RDS5, SGA5, REV5, EMPS5, and TEMP5 are ranked in an ascending manner, while CAP5 is ranked in a descending manner, so that a higher (lower) value of these variables reflects prospector (defender) characteristics. The range of scores for each type of business strategy is as follows: defender (6-12); analyzer (13-23); prospector (24-30). The six variables used to compute BOS include: (1) research intensity (RDS5) which is the ratio of research and development expenditures to sales computed over a rolling prior five-year average; (2) marketing and advertising efforts (SGA5) which is the ratio of selling, general and administrative expenses to sales computed over a rolling prior five-year average; (3) historical growth or investment opportunities (REV5) which is one-year percentage change in total sales computed over a rolling prior five-year average; (4) operational efficiency (EMPS5) which is the number of employees to sales computed over a rolling prior five-year average; (5) capital intensity or technological efficiency (CAP5) measured as net property, plant, and equipment to total assets computed over a rolling prior five-year average; and (6) organizational stability (TEMP5) which is the standard deviation of the total number of employees computed over a rolling prior five-year period. ABDA refers to the absolute value of Modified Jones (1991) discretionary accruals (ABDA) per Dechow et al. (1995). REAM is the sum of the standardized three real earnings management measures, namely abnormal cash flows from operations (RCFO), abnormal production costs (RPROD), and abnormal discretionary expense (RDISX), as in Cohen et al. (2008), Park (2017), and Roychowdhury (2006). EARN is operating earnings scaled by book value of assets. SIZE is the logarithm of market value of equity. MTB is the market value of equity plus book value of liability divided by the book value of total assets. AGE is the logarithm number of years since a firm shows up in CRSP monthly stock return files. SIAT is special items (SI) scaled by book value of assets. RETVOL is the standard deviation of the monthly stock returns in the last year. EVOL is the standard deviation of the operating earnings in the last five fiscal years. NBSEG is the logarithm of the number of business segments plus 1. NGSEG is the logarithm of the number of geographic segments plus 1. NITEMS is the logarithm of the number of non-missing items on Compustat. SEO is a dichotomous variable that equals 1 if a firm has seasoned equity offering during the year and 0 otherwise. MA is a dichotomous variable that equals 1 if a firm has merger and acquisition during the year and 0 otherwise. DLW is a dichotomous variable that equals 1 if a company is incorporated in Delaware and 0 otherwise. \*\*\*, \*\*, and <sup>\*</sup> indicate significance at 0.01, 0.05, and 0.10 level (two-tailed test), respectively.

10-Ks, and is measured as the number of Loughran-McDonald Financial-Litigious words in the document divided by the total number of words in the document that occur in the master dictionary. Litigious tone includes words such as *claimant, testimony, regulation* and *tort*, reflecting a propensity for legal contests or a litigious environment. Second, negative tone (*NEG*) refers to the Loughran-McDonald negative word proportion within 10-Ks, and is measured as the number of Loughran-McDonald Financial-Negative words in the document divided by the total number of words in the document that occur in the master dictionary. Negative tone

Correlations	Correlations and variance inflation factors.	inflation																
Variable	FOG	BOS	ABDA	REAM	EARN	SIZE	MTB	AGE	SIAT	RETVOL	EVOL	NBSEG	NGSEG	NITEMS	SEO	MA	DLW	VIF
FOG		0.046	0.026	0.037	-0.048	0.022	0.017	-0.034	-0.026	0.027	0.031	0.032	0.014	0.167	0.042	0.026	0.055	
		< .0001	<.0001	< .0001	< .0001	0.000	0.005	<.0001	< .0001	< .0001	< .0001	< .0001	0.022	< .0001	< .0001	< .0001	< .0001	
BOS	0.065		0.088	0.153	-0.072	0.176	0.238	-0.091	-0.040	0.105	0.127	-0.033	0.119	0.027	0.180	0.142	0.047	1.17
	< .0001		<.0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	
ABDA	0.044	0.054		0.152	-0.309	-0.201	0.110	-0.144	-0.089	0.282	0.286	-0.082	-0.034	-0.053	-0.032	-0.049	0.031	1.22
	< .0001	< .0001		< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	
REAM	0.061	0.181	0.092		-0.173	-0.010	0.138	-0.101	0.007	0.110	0.173	-0.049	0.051	0.047	0.055	0.028	0.054	1.09
	< .0001	< .0001	< .0001		< .0001	0.107	< .0001	< .0001	0.225	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	
EARN	-0.092	0.001	-0.167	-0.166		0.357	0.057	0.140	0.001	-0.341	-0.492	0.048	-0.000	0.000	0.111	0.120	-0.047	1.55
	< .0001	0.847	< .0001	< .0001		< .0001	< .0001	< .0001	0.809	< .0001	< .0001	< .0001	0.922	0.950	< .0001	< .0001	< .0001	
SIZE	0.006	0.175	-0.232	-0.020	0.442		0.302	0.263	-0.124	-0.351	-0.264	0.248	0.252	0.281	0.566	0.291	0.079	2.44
	0.343	< .0001	< .0001	0.001	< .0001		< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	
MTB	0.031	0.277	0.002	0.122	0.442	0.479		-0.092	0.002	0.051	0.192	-0.075	0.047	-0.008	0.217	-0.012	0.050	1.34
	< .0001	< .0001	0.701	< .0001	< .0001	< .0001		< .0001	0.647	< .0001	< .0001	< .0001	< .0001	0.159	< .0001	0.055	< .0001	
AGE	-0.047	-0.088	-0.139	-0.118	0.150	0.207	-0.055		-0.058	-0.254	-0.205	0.241	0.109	0.163	0.100	0.080	-0.146	1.23
	< .0001	< .0001	<.0001	< .0001	< .0001	< .0001	< .0001		< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	
SIAT	-0.053	-0.083	-0.040	-0.008	0.082	-0.204	0.031	-0.039		0.015	0.015	-0.078	-0.032	-0.062	-0.079	-0.038	-0.015	1.04
	< .0001	< .0001	<.0001	0.185	<.0001	< .0001	< .0001	<.0001		0.017	0.014	< .0001	< .0001	< .0001	< .0001	< .0001	0.013	
RETVOL	0.066	0.138	0.276	0.118	-0.348	-0.377	-0.094	-0.300	-0.029		0.362	-0.065	0.016	-0.076	-0.107	-0.147	0.051	1.38
	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001		< .0001	< .0001	0.009	< .0001	< .0001	< .0001	< .0001	
EVOL	0.045	0.109	0.249	0.144	-0.273	-0.338	0.073	-0.240	0.075	0.434		-0.126	-0.028	-0.072	-0.070	-0.140	0.069	1.55
	< .0001	< .0001	<.0001	< .0001	<.0001	< .0001	<.0001	<.0001	< .0001	< .0001		< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	
NBSEG	0.031	-0.033	-0.084	-0.043	-0.022	0.210	-0.059	0.176	-0.135	-0.040	-0.169		0.359	0.300	0.127	0.195	0.006	1.30
	< .0001	< .0001	< .0001	< .0001	0.000	< .0001	< .0001	<.0001	< .0001	< .0001	< .0001		< .0001	< .0001	< .0001	< .0001	0.286	
NGSEG	0.038	0.121	-0.045	0.076	-0.041	0.270	0.092	0.081	-0.183	0.034	-0.047	0.490		0.431	0.180	0.133	0.044	1.37
	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001		< .0001	< .0001	< .0001	< .0001	
NITEMS	0.182	0.019	-0.066	0.060	-0.029	0.296	0.072	0.143	-0.213	-0.069	-0.124	0.381	0.513		0.184	0.105	0.070	1.33
	< .0001	0.001	<.0001	< .0001	<.0001	< .0001	<.0001	<.0001	< .0001	< .0001	< .0001	< .0001	< .0001		< .0001	< .0001	< .0001	
SEO	0.047	0.182	-0.062	0.057	0.184	0.568	0.314	0.072	-0.155	-0.145	-0.125	0.116	0.195	0.188		0.169	0.091	1.53
	< .0001	< .0001	< .0001	< .0001	<.0001	< .0001	<.0001	<.0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001		< .0001	< .0001	
MA	0.029	0.138	-0.078	0.023	0.127	0.298	0.067	0.060	-0.140	-0.156	-0.198	0.186	0.153	0.106	0.169		0.017	1.15
	< .0001	< .0001	< .0001	0.000	<.0001	< .0001	<.0001	<.0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001	< .0001		0.006	
DLW	0.066	0.048	0.012	0.063	-0.048	0.083	0.052	-0.154	-0.057	0.065	0.067	0.005	0.051	0.071	0.091	0.017		1.05
	< .0001	< .0001	0.049	<.0001	< .0001	<.0001	< .0001	<.0001	< .0001	< .0001	< .0001	0.403	< .0001	< .0001	< .0001	0.006		
Dearcon (Sn	1000. >	<.0001 <.0001 0.049 <.0001 <.0001 <.0001 <.0001 <.0001 <.0001 <.0001 <.0001 <.0001 <.0001 <.0001 0.403 <.0001 <.0001 <.0001 <.0005	0.049 renorted in	<. 0001 <	< .0001	<ul> <li>&lt; .0001</li> <li></li> <li></li></ul>	< .0001	< .0001	< .0001	>	<ul> <li>&lt; .0001</li> <li></li> <li></li></ul>	0.403 in hold fe	<ul> <li>&lt; .0001</li> <li></li> <li></li></ul>	< .0001	< .0001	0.006		c
variable definitions			reputeu III	nie uppei (it	WELD MAROL	1al. 111C p-Va.										Cintion to		

#### Table 4

Regressions of FOG on business strategy and control variables.

		Full sample		PRO and DEF only
	Exp sign	(1)	(2)	(3)
Intercept	?	18.310	18.507	14.920
		$(11.28)^{***}$	(11.35)***	(3.99)***
BOS	+	0.016		
		(3.64)****		
DEF	-		-0.125	
			(-2.37)**	
PRO	+		0.186	0.324
			$(3.13)^{***}$	(3.35)***
LAG_FOG	+	0.178	0.178	0.151
		(17.13)***	(17.16)***	$(7.22)^{***}$
ABDA	+	0.048	0.043	0.472
		(0.35)	(0.32)	$(1.73)^{*}$
REAM	+	0.015	0.025	-0.252
		(0.22)	(0.38)	(-1.92)*
EARN	-	-0.243	-0.242	-0.283
		(-2.97)***	(-2.97)****	$(-1.71)^*$
SIZE	+	0.010	0.012	0.039
		(0.89)	(1.07)	(1.35)
MTB	+	0.001	0.003	-0.000
		(0.18)	(0.38)	(-0.02)
AGE	_	-0.051	-0.052	-0.049
		(-1.99)**	(-2.03)**	(-0.74)
SIAT	_	0.012	0.013	0.150
		(0.36)	(0.39)	(1.24)
RETVOL	+	0.327	0.357	0.486
		(2.26)**	(2.46)**	(1.26)
EVOL	+	-0.020	0.012	-0.415
		(-0.10)	(0.06)	(-0.99)
NBSEG	+	0.022	0.016	-0.103
		(0.41)	(0.30)	(-0.72)
NGSEG	+	-0.288	-0.276	-0.287
		(-5.01)***	(-4.82)***	$(-1.93)^*$
NITEMS	+	-0.894	-0.873	0.866
		(-1.40)	(-1.37)	(0.58)
SEO	+	0.059	0.062	-0.035
		(1.63)*	(1.70)*	(-0.34)
MA	+	0.013	0.018	0.081
		(0.49)	(0.65)	(1.05)
DLW	+/-	0.094	0.094	0.075
	• /	(2.81)***	(2.79)***	(0.89)
PRO = DEF F-statistic		(2.01)	13.23	(0.05)
(p-value)			(0.000)***	
(p-value) Adj. R-square		11.98%	11.98%	11.28%
Nuj. K-square		24,817	24,817	3157
11		24,017	24,017	313/

DEF is equal to 1 for defenders, and 0 otherwise. PRO is equal to 1 for prospectors, and 0 otherwise. LAG\_FOG is FOG lagged at *t*-1. See Table 2 for other variable definitions. Reported results include industry and year fixed effects in regressions, with robust standard errors clustered by firm. The *t*-statistics are in parentheses. \*\*\*, \*\*\*, and \* indicate significance at 0.01, 0.05, and 0.10 level (two-tailed test), respectively.

incorporates words such as *loss, losses, adverse* and *failure*, which indicate pessimism. Third, uncertainty tone (*UNC*) refers to the Loughran-McDonald uncertainty word proportion within 10-Ks, which is the number of Loughran-McDonald Financial-Uncertainty words in the document divided by the total number of words in the document that occur in the master dictionary. Uncertainty tone comprises words such as *approximate, contingency, depend, fluctuate* and *variability*, which signify imprecision.

To examine the association between disclosure tone and business strategy, we separately regress each measure of tone (*LIT*, *NEG*, and *UNC*) on three distinct test variable specifications: (1) *BOS*; (2) *PRO* (*DEF*), which is a dichotomous variable that equals 1 for prospectors (defenders) and 0 otherwise; and (3) *PRO*, which is a dichotomous variable that equals 1 for prospectors and 0 otherwise; after limiting the sample to prospectors and defenders only. We employ a similar set of controls to those used in Model 1.<sup>10</sup>

The results for this analysis are reported in Table 5. With regard to *LIT*, column (1) of Table 5 reports a negative coefficient on *BOS* (-0.012) that is significant at the one percent level (*t*-statistic = -5.07). Upon replacing *BOS* with the separate dichotomous variables *DEF* and *PRO*, column (2) shows a significant positive (negative) coefficient on *DEF* (*PRO*) at the five (ten) percent level, indicating

<sup>&</sup>lt;sup>10</sup> Although we do not control for lagged disclosure tones in this analysis, consistent results (untabulated) are observed after further controlling for lagged disclosure tones in the model.

#### Table 5

Regressions of disclosure sentiment on business strategy and control variables.

	Full sample							PRO and DEF only			
	DV = LIT		DV = NEG		DV = UNC		DV = LIT	DV = NEG	DV = UNC		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)		
Intercept	-0.168 (-0.19)	-0.273 (-0.31)	-2.541 $(-4.35)^{***}$	-2.479 (-4.24) <sup>***</sup>	1.382 (3.70) <sup>***</sup>	1.420 (3.78) <sup>***</sup>	1.453 (0.75)	-3.685 $(-2.72)^{***}$	-0.985 (-1.12)		
BOS	$-0.012$ $(-5.07)^{***}$		0.010 (5.99) <sup>****</sup>		0.010 (9.05) <sup>***</sup>						
DEF		0.069 (2.35) <sup>**</sup>		-0.048 $(-2.58)^{***}$		-0.035 $(-2.57)^{***}$					
PRO		-0.042 $(-1.62)^*$		0.085 (3.62)***		0.070 (4.23) <sup>***</sup>	(-0.130)	0.164 (4.97) <sup>***</sup>	0.123 $(5.57)^{***}$		
ABDA	-0.002	-0.008	-0.043	-0.043	0.002	0.004	-0.147	-0.084	-0.045		
	(-0.04)	(-0.12)	(-0.86)	(-0.88)	(0.06)	(0.12)	(-1.15)	(-0.83)	(-0.58)		
REAM	-0.028	-0.040	0.103	0.112	0.029	0.041	-0.012	0.057	-0.003		
	(-0.92)	(-1.33)	(4.61)***	(4.98)***	(2.15)**	(2.92)***	(-0.18)	(1.23)	(-0.10)		
EARN	-0.054	-0.041	-0.222	-0.226	-0.047	-0.053	-0.087	-0.079	-0.026		
	(-1.55)	(-1.20)	$(-4.81)^{***}$	(-4.87)***	$(-2.25)^{**}$	$(-2.55)^{***}$	(-1.15)	(-1.28)	(-0.59)		
SIZE	0.037	0.035	0.019	0.021	0.008	0.009	0.018	0.001	0.004		
	(5.72)***	(5.42)***	(4.16)***	(4.49)***	$(2.67)^{***}$	$(3.21)^{***}$	(1.19)	(0.13)	(0.57)		
МТВ	-0.028	-0.025	-0.027	-0.025	0.000	0.002	-0.015	-0.016	0.005		
	(-5.66)***	(-6.22)***	(-6.34)***	(-5.98)***	(0.00)	(0.87)	(-1.96)**	$(-2.38)^{**}$	(0.91)		
AGE	0.140	0.142	-0.021	-0.022	-0.089	-0.089	0.135	-0.015	-0.064		
	(9.72)***	(9.86)***	$(-2.10)^{**}$	$(-2.17)^{**}$	$(-12.74)^{***}$	(-12.65)***	(4.13)***	(-0.72)	(-4.03)***		
SIAT	-0.054	-0.055	-0.078	-0.077	-0.005	-0.003	-0.004	0.005	-0.009		
	(-1.65)*	(-1.59)	$(-2.27)^{**}$	$(-2.33)^{**}$	(-0.50)	(-0.46)	(-0.09)	(0.14)	(-0.32)		
RETVOL	-0.055	-0.083	0.677	0.700	0.157	0.183	-0.197	0.641	0.203		
	(-0.81)	(-1.22)	(12.03)***	(12.44)***	(4.15)***	(4.79)***	(-1.19)	(5.19)***	$(2.30)^{**}$		
EVOL	0.007	-0.029	0.482	0.507	0.095	0.123	-0.065	0.369	0.095		
	(0.07)	(-0.31)	(5.34)***	(5.60)***	(1.90)*	(2.45)**	(-0.33)	(2.09)**	(0.85)		
NBSEG	0.082	0.088	0.005	0.000	-0.111	-0.117	0.038	-0.031	-0.105		
	(2.88)***	$(3.11)^{***}$	(0.25)	(0.00)	(-7.80)***	(-8.20)***	(0.60)	(-0.64)	(-2.75)***		
NGSEG	-0.074	-0.086	0.066	0.077	0.046	0.058	-0.026	0.127	0.066		
	(-2.49)**	(-2.90)***	(2.83)***	(3.28)***	(2.89)***	(3.68)***	(-0.40)	$(2.65)^{***}$	$(1.79)^{*}$		
NITEMS	0.238	0.209	1.547	1.583	0.083	0.129	-0.441	1.964	0.992		
	(0.68)	(0.60)	(6.57)***	(6.71)***	(0.55)	(0.85)	(-0.57)	(3.61)***	$(2.82)^{***}$		
SEO	0.015	0.012	0.037	0.040	0.018	0.022	0.026	0.013	-0.003		
	(0.85)	(0.61)	(2.98)***	$(3.20)^{***}$	(2.33)**	(2.75)***	(0.67)	(0.49)	(-0.17)		
MA	0.001	-0.006	-0.040	-0.035	0.001	0.007	0.018	-0.017	0.006		
	(0.05)	(-0.41)	(-3.99)***	(-3.50)***	(0.21)	(1.20)	(0.54)	(-0.71)	(0.38)		
DLW	0.037	0.037	0.056	0.058	0.025	0.025	0.114	0.076	0.030		
	$(1.91)^{*}$	$(1.91)^{*}$	(4.30)***	(4.25)***	$(2.77)^{***}$	(2.69)***	(2.57)***	(2.68)***	(1.28)		
PRO = DEF											
F-statistic		9.89		25.32		42.22					
(p-value)		(0.000)***		(0.000)***		(0.000)***					
Adj. R-square	9.62%	9.36%	30.68%	30.45%	47.36%	46.70%	9.55%	33.02%	4.44%		
Ν	18,105	18,105	18,105	18,105	18,105	18,105	2312	2312	2312		

LIT refers to the Loughran-McDonald litigious word proportion in the 10-K filings, which is the number of Loughran-McDonald Financial-Litigious words in the document divided by the total number of words in the document that occur in the master dictionary. NEG refers to the Loughran-McDonald negative word proportion in the 10-K filings, which is the number of Loughran-McDonald Financial-Negative words in the document divided by the total number of words in the document that occur in the master dictionary. UNC refers to the Loughran-McDonald uncertainty word proportion in the 10-K filings, which is the number of Loughran-McDonald uncertainty word proportion in the 10-K filings, which is the number of Loughran-McDonald Financial-Negative words in the document divided by the total number of Loughran-McDonald Financial-Uncertainty words in the document divided by the total number of words in the document that occur in the master dictionary. See Tables 2 and 4 for other variable definitions. Reported results include industry and year fixed effects in regressions, with robust standard errors clustered by firm. The *t*-statistics are in parentheses. \*\*\*, \*\*, and \* indicate significance at 0.01, 0.05, and 0.10 level (two-tailed test), respectively.

that defenders (prospectors) have more (less) litigious tone in their 10-Ks than their non-counterparts. Upon excluding analyzers, column (7) reports a negative coefficient on *PRO* (-0.130) that is significant at the one percent level (*t*-statistic = -3.18). Overall, our findings suggest that defenders incorporate more litigious tone in their 10-Ks than prospectors. A plausible explanation for this is that given the risk averse nature of defenders they tend to comply more with regulation to avoid reputation costs (Higgins et al., 2015), and risk averse firms tend to employ more litigious tones to inform the market of the firm's legal environment so as to mitigate litigation risks (Martikainen et al., 2016; Rogers et al., 2011).

With regard to *NEG*, column (3) reports a significant positive coefficient on *BOS* (0.010, *t*-statistic = 5.99), while column (4) shows a significant negative (positive) coefficient on *DEF* (*PRO*), with both *t*-statistics significant at the one percent level. When limiting the sample to prospectors and defenders only, column (8) reports a positive coefficient on *PRO* (0.164) that is significant at the one percent level (*t*-statistic = 4.97). Overall, these findings are consistent with prospectors including more negative tone in their

#### E.K. Lim et al.

#### Journal of Accounting and Public Policy xxx (xxxx) xxx-xxx

10-Ks than both analyzers and defenders. Finally, Table 5 indicates that prospectors tend to also include more uncertain tone in their 10-Ks than other firms, given the significant positive coefficient on *BOS* in column (5), the significant negative (positive) coefficient on *DEF (PRO)* in column (6), and the significant positive coefficient on *PRO* in column (9). Across model specifications, columns (2), (4) and (6) also report that, upon undertaking an *F*-test of coefficient equality, the *DEF* and *PRO* coefficients consistently are not equal. Collectively, the findings for *NEG* and *UNC* reflect prospectors' 10-Ks displaying more negative and uncertain tones, which is consistent with these firms operating in volatile and risky environments that can beget less readable 10-Ks stemming from complex explanations for uncertain or unsuccessful projects. The evidence bridges the insights provided by several prior studies, including Naiker et al. (2008), who show that prospectors have greater stock return volatility and uncertainty relative to defenders, while Kothari et al. (2009) associate negativity in management's disclosures with a firm's stock return volatility.

#### 5.2. Alternative measures of 10-K readability

Following Li (2008), we adopt three alternative measures of annual report readability: (1) the Kincaid Index, also known as the -Flesh-Kincaid Grade Level (*KINCAID*), where a higher value of *KINCAID* indicates lower readability<sup>11</sup>; (2) the Flesch Reading Ease (*FLESCH*), where a higher value of *FLESCH* signifies greater readability<sup>12</sup>; and (3) the length of an annual report (*LENGTH*), which is estimated as the logarithm of the number of words in a 10-K, where a higher value of *LENGTH* reflects lower readability. These data are also obtained from Feng Li's website. The results for this analysis are reported in Table 6.

Column (1) shows a positive coefficient on *BOS* (0.021) using *KINCAID* as the alternative 10-K readability proxy, while column (3) shows a negative coefficient on *BOS* (-0.098) employing *FLESCH* as the alternative 10-K readability proxy, with both significant at the one percent level (*t*-statistics = 5.19 and -8.28, respectively). This is consistent with prospectors' 10-Ks containing more complex information relative to defenders and analyzers, which requires greater user sophistication to understand and interpret. In support, when using *KINCAID* to measure readability, column (2) shows a significant negative (positive) coefficient on *DEF* (*PRO*), with both *t*-statistics significant at the one percent level, while column (4) shows a significant positive (negative) coefficient on *DEF* (*PRO*) at the one percent level when using *FLESCH* as the dependent variable. An *F*-test of coefficient equality shows that the *DEF* and *PRO* coefficients are not equal, whether in the context of *KINCAID* (*F*-statistic = 17.80, *p*-value = 0.000) or *FLESCH* (*F*-statistic = 39.19, *p*-value = 0.000). Moreover, when limiting the sample to prospectors and defenders only, the significant positive association between *KINCAID* and *PRO* (-1.501, *t*-statistic = -5.77), indicate prospectors have less readable 10-Ks than defenders. We do, however, fail to find consistent evidence that business strategy accounts for variation in 10-K length, which aligns with Cazier and Pfeiffer (2016) who fail to find an association between R&D expenditure, which is one aspect of business strategy, and 10-K length. This also highlights that *FOG*, *KINCAID*, and *FLESCH* capture different facets of readability from *LENGTH*.

#### 5.3. Earnings performance

We examine whether the association between business strategy and 10-K readability is sensitive to firm performance, given Li (2008) finds that poorly performing firms have less readable annual reports. We re-estimate Model 1 based on two sub-samples of firms depending on whether their current period earnings (*EARN*) is above or below industry-year median *EARN* to examine whether our main findings hold across both sub-samples.

Results reported in Table 7 show a significant positive coefficient on *BOS*, whether *EARN* is above (0.016, *t*-statistic = 2.50) or below (0.017, *t*-statistic = 3.18) the industry-year median. Moreover, we continue to find a significant positive coefficient on *PRO* across both *EARN* sub-samples, and a significant negative coefficient on *DEF* for the sub-sample where *EARN* is below the industry-year median. *F*-tests indicate that the *DEF* and *PRO* coefficients are not equal for both the above (*F*-statistic = 4.16, *p*-value = 0.016) and below (*F*-statistic = 8.95, *p*-value = 0.000) industry-year median *EARN* sub-samples. When limiting the sample to defenders and prospectors, a significant positive coefficient on *PRO* is found for both the above (0.483, *t*-statistic = 3.22) and below (0.250, *t*-statistic = 2.17) *EARN* sub-samples. In sum, our main findings are not sensitive to firm performance suggesting that business strategy fundamentally determines *FOG* for both good and poor performing firms.<sup>13</sup>

#### 6. Robustness tests

Potential endogeneity issues are addressed with the following tests.

 $<sup>^{11}</sup>$  KINCAID, indicative of a document that is understandable by an average US grade level, is estimated as (11.8 \* number of syllables/number of words) + (0.39 \* number of words/number of sentences) - 15.59. For example, a score of 8.0 indicates a document that is understandable by an average eighth grader.

<sup>&</sup>lt;sup>12</sup> FLESCH rates reading ease on a scale of 0–100, where a score of 0 (1 0 0) reflects the material hardest (easiest) to read, calculated as 206.835 \* (1.015 \* number of words/number of sentences) – (84.6 \* number of syllables/number of words).

<sup>&</sup>lt;sup>13</sup> We also re-estimate this analysis within the sub-sample of firms with improved earnings performance from *t*-1 to *t*, and within the sub-sample of firms with deteriorated earnings performance from *t*-1 to *t*. We continue to observe that *BOS* significantly explains the variation in *FOG*, and prospectors have greater *FOG* relative to defenders, within each sub-sample of improved and deteriorated earnings performance.

#### E.K. Lim et al.

#### Table 6

Regressions of alternative readability measures on business strategy and control variables.

	Full sample				PRO and DEF only				
	DV = KINC	AID	DV = FLESC	ЭН	DV = LENG	TH	DV = KINCAID	DV = FLESCH	DV = LENGTH
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Intercept	14.035 (9.24) <sup>***</sup>	14.221 (9.31) <sup>***</sup>	16.365 (4.01) <sup>***</sup>	15.202 (3.71) <sup>***</sup>	5.603 (8.93) <sup>***</sup>	5.555 (8.87) <sup>***</sup>	8.368 (2.39) <sup>**</sup>	31.720 (3.02) <sup>***</sup>	5.129 (3.75) <sup>****</sup>
BOS	0.021 (5.19) <sup>***</sup>		(-0.098) $(-8.28)^{***}$		0.002 (1.37)				
DEF		-0.131 $(-2.64)^{***}$		0.697 (4.85) <sup>***</sup>		0.021 (1.06)			
PRO		0.219 (3.83) <sup>***</sup>		-0.744 $(-4.58)^{***}$		0.084 (3.92) <sup>***</sup>	0.383 (4.19) <sup>***</sup>	(-1.501)	0.044 (1.37)
LAG_READ	0.212 (20.63) <sup>***</sup>	0.213 (20.68) <sup>***</sup>	$0.309$ $(27.40)^{***}$	0.312 (27.56) <sup>***</sup>	0.264 (22.39) <sup>****</sup>	0.264 (22.36)***	0.183 (8.53) <sup>***</sup>	0.305 (11.29) <sup>***</sup>	0.250 (9.74) <sup>***</sup>
ABDA	0.042	0.040	0.694 (1.81) <sup>*</sup>	0.682 (1.78) <sup>*</sup>	0.200 (3.24) <sup>***</sup>	0.196 (3.17) <sup>***</sup>	0.469 (1.81) <sup>*</sup>	(-1.006) (-1.34)	0.275 (2.27) <sup>**</sup>
REAM	0.064 (1.03)	0.081 (1.30)	(-0.019) (-0.10)	(-0.100) (-0.51)	0.012 (0.43)	0.013 (0.48)	(-0.287) $(-2.37)^{**}$	0.525 (1.54)	(2.27) - 0.087 (-1.34)
EARN	(-0.312) $(-3.88)^{***}$	$(-3.90)^{***}$	(-0.10) 1.036 (4.38) <sup>***</sup>	(-0.31) 1.077 (4.54)***	$(-8.33)^{***}$	$(-8.23)^{***}$	(-2.37) -0.426 $(-2.88)^{***}$	(1.34) 1.135 (2.54) <sup>**</sup>	(-1.34) -0.352 $(-5.23)^{***}$
SIZE	0.043	0.046	-0.158	-0.173	0.089	0.089	0.070	-0.238	(-5.23) 0.103 (9.11) <sup>***</sup>
МТВ	(3.85) <sup>***</sup> -0.004	(4.12) <sup>***</sup> -0.001	(-5.17) <sup>***</sup> 0.004	(-5.63) <sup>***</sup> -0.012	(18.19) <sup>****</sup> -0.040	(18.27) <sup>****</sup> -0.040	$(2.53)^{**}$ - 0.008	(-3.07) <sup>***</sup> 0.042	-0.024
AGE	(-0.54) -0.058	(-0.13) -0.059	(0.19) 0.117	(-0.50) 0.127	(-9.58)*** -0.076	(-9.66)*** -0.074	(-0.50) -0.075	(0.87) 0.160	$(-3.74)^{***}$ -0.061
SIAT	(-2.36) <sup>**</sup> 0.006	$(-2.41)^{**}$ 0.008	$(1.72)^{*}$ 0.027	$(1.87)^{*}$ 0.017	$(-7.36)^{***}$ -0.047	$(-7.20)^{***}$ -0.047	(-1.19) 0.213	(0.88) - 0.697	(-2.37)** 0.122
RETVOL	(0.19) 0.352	(0.25) 0.398	(0.22) - 0.859	(0.14) -1.066	(-2.36) <sup>**</sup> 0.529	( – 2.35) <sup>**</sup> 0.535	(1.99) <sup>*</sup> 0.500	$(-2.06)^{**}$ -1.654	(1.83) <sup>*</sup> 0.694
EVOL	(2.57) <sup>***</sup> 0.091	(2.91) <sup>***</sup> 0.139	(-2.18) <sup>**</sup> 1.209	(-2.70) <sup>***</sup> 0.971	(7.65) <sup>***</sup> 0.012	(7.76) <sup>***</sup> 0.012	(1.36) -0.542	( – 1.65) <sup>*</sup> 2.548	(4.34) <sup>***</sup> 0.025
NBSEG	(0.46) 0.022	(0.71) 0.012	(1.94) <sup>*</sup> - 0.181	(1.56) -0.137	(0.12) 0.113	(0.12) 0.112	(-1.30) -0.096	(2.10) <sup>**</sup> 0.327	(0.16) 0.123
NGSEG	(0.42) - 0.230	(0.24) -0.210	(-1.26) 0.255	(-0.96) 0.170	(5.22) <sup>***</sup> 0.039	(5.17) <sup>***</sup> 0.044	(-0.69) -0.219	(0.84) 0.142	(2.36) <sup>**</sup> -0.001
NITEMS	$(-4.19)^{***}$ -0.757	$(-3.84)^{***}$ -0.705	(1.67) <sup>*</sup> 0.454	(1.11) 0.295	$(1.67)^*$ 0.369	(1.86) <sup>*</sup> 0.401	(-1.57) 1.911	(0.36) - 5.982	(-0.03) 0.696
SEO	(-1.26) 0.059	(-1.17) 0.064	(0.28) 0.023	(0.18) -0.002	(1.51) 0.061	(1.65) <sup>*</sup> 0.060	(1.36) -0.033	(-1.45) 0.297	(1.34) 0.054
	$(1.68)^{*}$	(1.82)*	(0.24)	(-0.02)	(3.98)***	(3.97)***	(-0.33)	(1.08)	(1.53)
MA	0.007 (0.27)	0.015 (0.60)	-0.058 (-0.79)	-0.097 (-1.33)	0.038 (3.29) <sup>***</sup>	0.039 (3.42) <sup>***</sup>	0.060 (0.79)	-0.291 (-1.39)	0.037 (1.29)
DLW	$0.089$ $(2.79)^{***}$	0.088 $(2.76)^{***}$	-0.181 $(-2.05)^{**}$	$-0.179$ $(-2.02)^{**}$	0.079 (5.85) <sup>***</sup>	0.078 $(5.81)^{***}$	0.100 (1.25)	-0.130 (-0.57)	0.128 (4.26) <sup>***</sup>
PRO = DEF F-statistic		17.80		39.19		7.56			
(p-value)		(0.000)***		(0.000)***		(0.000)***			
Adj. <i>R</i> -square <i>N</i>	15.20% 24,817	15.14% 24,817	25.73% 24,817	25.56% 24,817	20.11% 24,817	20.14% 24,817	15.69% 3157	27.04% 3157	25.12% 3157

KINCAID refers to the Kincaid Index, which estimates the understandability of a 10-K by an average US grade level, and is measured as  $(11.8 \times number of syllables/number of words) + (0.39 \times number of words/number of sentences) - 15.59$ . FLESCH refers to the reading ease of a 10-K on a scale of 0-100, and is calculated as  $206.835 \times (1.015 \times number of words/number of sentences) - (84.6 \times number of syllables/number of words)$ . LENGTH refers to the length of a 10-K, and is measured as the logarithm of the number of words in a 10-K. LAG\_READ is KINCAID, FLESCH and LENGTH lagged at t-1 when KINCAID, FLESCH and LENGTH is used as the dependent variable, respectively. See Tables 2 and 4 for other variable definitions. Reported results include industry and year fixed effects in regressions, with robust standard errors clustered by firm. The t-statistics are in parentheses. \*\*\*, \*\*\*, and \* indicate significance at 0.01, 0.05, and 0.10 level (two-tailed test), respectively.

#### 6.1. Lead-lag regression

In line with Fich and Shivdasani (2006), we adopt a lead-lag approach to mitigate the concern of reverse causality. To do this, each independent variable is lagged by one year. In untabulated findings, for the full sample the coefficient on *LAG\_DEF* is negative (-0.126) and significant (*t*-statistic = -2.42) while the coefficient on *LAG\_PRO* is positive (0.114) and significant (*t*-statistic = 1.85).<sup>14</sup> In addition, an *F*-test indicates that the coefficients on *LAG\_DEF* and *LAG\_PRO* are not equal (*F*-statistic = 7.22, *p*-

<sup>&</sup>lt;sup>14</sup> Untabulated results are available upon request.

#### Table 7

Earnings performance, business strategy, and 10-K readability.

	Above industry-	year earnings median		Below industry-year earnings median			
	Full sample		DEF and PRO only	Full sample		DEF and PRO only	
	(1)	(2)	(3)	(1)	(2)	(3)	
Intercept	20.669	21.023	24.397	14.778	15.023	5.081	
	(9.03)***	(9.12)***	(4.10)***	(7.37)***	(7.44)***	(1.04)	
BOS	0.016			0.017			
	(2.50)**			(3.18)****			
DEF		-0.121			-0.158		
		(-1.57)			$(-2.47)^{**}$		
PRO		0.166	0.483		0.155	0.250	
		(1.69)*	(3.22)***		(2.30)**	$(2.17)^{**}$	
LAG_FOG	0.192	0.192	0.168	0.160	0.160	0.141	
	(11.72)***	$(11.71)^{***}$	(5.32)***	$(12.52)^{***}$	$(12.53)^{***}$	(5.05)***	
ABDA	0.644	0.648	0.430	-0.216	-0.221	0.368	
	$(2.61)^{***}$	(2.62)****	(0.59)	(-1.36)	(-1.39)	(1.22)	
REAM	-0.046	-0.035	-0.463	0.016	0.026	-0.132	
	(-0.49)	(-0.38)	$(-2.29)^{**}$	(0.18)	(0.30)	(-0.70)	
EARN	-1.159	-1.150	-1.211	-0.095	-0.096	-0.099	
	(-3.86)***	(-3.80)***	(-1.32)	(-1.03)	(-1.05)	(-0.54)	
SIZE	0.009	0.013	0.033	0.028	0.030	0.047	
	(0.51)	(0.79)	(0.63)	(2.04)**	$(2.17)^{**}$	(1.51)	
MTB	0.025	0.027	-0.003	0.016	0.018	0.039	
	(1.86)*	(1.94)*	(-0.10)	(1.57)	$(1.71)^{*}$	(1.60)	
AGE	-0.013	-0.014	-0.068	-0.075	-0.077	-0.022	
	(-0.38)	(-0.42)	(-0.74)	$(-2.25)^{**}$	$(-2.33)^{**}$	(-0.28)	
SIAT	0.085	0.088	-0.372	-0.051	-0.055	0.157	
	(2.55)**	(2.62)****	(-0.36)	(-0.72)	(-0.78)	(1.30)	
RETVOL	0.126	0.191	0.568	0.322	0.335	0.620	
	(0.46)	(0.71)	(0.70)	(1.95)*	(2.04)**	(1.51)	
EVOL	0.839	0.869	- 0.857	-0.050	-0.006	-0.168	
	(2.16)**	(2.23)**	(-0.97)	(-0.2)	(-0.03)	(-0.35)	
NBSEG	0.000	-0.009	0.000	0.002	-0.003	-0.217	
	(0.01)	(-0.12)	(0.00)	(0.04)	(-0.04)	(-1.31)	
NGSEG	-0.304	-0.299	-0.487	-0.311	-0.300	-0.247	
	(-3.90)***	(-3.84)***	$(-2.16)^{**}$	(-4.27)***	(-4.15)***	(-1.44)	
NITEMS	-2.034	-2.080	- 2.587	0.727	0.738	4.540	
	$(-2.29)^{**}$	(-2.35)**	(-1.09)	(0.91)	(0.92)	(2.35)**	
SEO	0.080	0.083	0.100	0.034	0.035	-0.145	
	(1.65)*	$(1.71)^*$	(0.59)	(0.66)	(0.69)	(-1.17)	
MA	-0.008	-0.001	-0.154	0.036	0.039	0.216	
WIA .	(-0.22)	(-0.03)	(-1.28)	(0.99)	(1.08)	(2.34)**	
DLW	0.126	0.130	0.183	0.042	0.042	0.004	
2111	(2.91)***	(2.99)***	(1.62)	(0.98)	(0.98)	(0.04)	
PRO = DEF	(4.71)	(2.77)	(1.02)	(0.90)	(0.50)	(0.04)	
F-statistic		4.16			8.95		
					8.95 (0.000) <sup>***</sup>		
(p-value)	10 100/	(0.016)**	12.940/	10.05%		11.000/	
Adj. <i>R</i> -square	13.13%	13.13%	13.84%	10.95%	10.95%	11.09%	
Ν	12,415	12,415	1196	12,402	12,402	1961	

Firms are split into sub-samples depending on whether their current period earnings (*EARN*) is above or below industry-year median *EARN*. See Tables 2 and 4 for other variable definitions. Reported results include industry and year fixed effects in regressions, with robust standard errors clustered by firm. The *t*-statistics are in parentheses. \*\*\*, \*\*, and \* indicate significance at 0.01, 0.05, and 0.10 level (two-tailed test), respectively.

value = 0.000). Upon eliminating analyzers, there is a significant positive coefficient on  $LAG_PRO$  (0.193, *t*-statistic = 2.06), which indicates our main finding that prospectors have less readable 10-Ks than defenders is not attributable to reverse causality.

#### 6.2. Firm age

Following Higgins et al. (2015), firm age analysis is conducted to discount the possibility that our results are driven by firm age. We do this by interacting a firm age indicator (*OLDER*) with *PRO* and *DEF* for the full sample, and with *PRO* when limiting the sample to prospectors and defenders only. *OLDER* equals 1 if the firm's age (*AGE*) is greater than the median industry-year age, and 0 otherwise. If age differences between prospectors and defenders are driving the main findings, each interaction term should be significant and the difference between the coefficients on *PRO* and *DEF* should be insignificant for the full sample analysis.

For the full sample of firm-years, our untabulated findings show the coefficients on both interaction terms (*DEF* \* *OLDER* and *PRO* \* *OLDER*) are insignificant, while the coefficient on *DEF* remains negative (-0.128) and significant (*t*-statistic = -1.86) and the

#### E.K. Lim et al.

#### Journal of Accounting and Public Policy xxx (xxxx) xxx-xxx

coefficient on *PRO* remains positive (0.185) and significant (*t*-statistic = 2.68). Moreover, an *F*-test of the equality of the *DEF* and *PRO* coefficients rejects the null hypothesis that these coefficients are equal (*F*-statistic = 8.48, *p*-value = 0.000). Upon excluding analyzers, there is an insignificant coefficient on *PRO*  $\star$  *OLDER* and a significant positive coefficient on *PRO* (0.339, *t*-statistic = 3.32). Overall, the firm age analysis results are consistent with the main findings and suggests that business strategy and firm age are different constructs.

#### 6.3. Firm size

To discount that firm size is driving our inferences, following Bentley et al. (2013) and Carson and Fargher (2007) we form size quintiles and expand the main model by interacting *PRO* and *DEF* with the size quintiles representing the smallest, median, and largest firms in the sample (i.e. *DEF* \* *SIZEQ1*, *DEF* \* *SIZEQ3*, *DEF* \* *SIZEQ5*, *PRO* \* *SIZEQ1*, *PRO* \* *SIZEQ3*, *PRO* \* *SIZEQ5*). To the extent a particular size group is driving the main findings, a significant coefficient on the respective interactive variable would result.

Untabulated results for the full sample show that all of the interactive variables have insignificant coefficients. Moreover, the results continue to show a significant negative coefficient on *DEF* (-0.133, *t*-statistic = -1.76) and a significant positive coefficient on *PRO* (0.174, *t*-statistic = 2.15), with an *F*-test indicating the *DEF* and *PRO* coefficients are not equal (*F*-statistic = 5.79, *p*-value = 0.000). After limiting the sample to prospectors and defenders, the interactive variables remain insignificant while *PRO* has a positive coefficient (0.322) significant at the one percent level (*t*-statistic = 3.04). In sum, these results indicate that firm size does not drive our main findings.

#### 7. Conclusion

Our study investigates whether business strategy is associated with 10-K readability. Consistent with the hypothesis, we find that prospectors have less readable 10-Ks relative to defenders. The results are robust to several sensitivity and robustness checks. We also find that prospectors' 10-Ks display more negative and uncertain tones, while defenders' 10-Ks exhibit more litigious tone. Our study suggests that the influence of business strategy on a firm's operating complexity and environmental uncertainty (and, hence, managerial opportunism) frames the level, wording, and complexity of disclosures and eventually 10-K readability.

Policy makers, locally and globally, are addressing calls for more streamlined reports with sources of streamlining including a plain English approach, reconsideration of materiality, grouping and reordering of note disclosures, reconfiguration of accounting policy disclosures and reduced page length. Our study is insightful for informing these deliberations. It suggests that policy makers need to be cognizant of the influence of a firm's strategic orientation on shaping the complexity and narrative of its communications. The success of regulatory initiatives to improve and monitor the readability of annual reports will vary depending on firms' strategic orientation. By necessity, prospector-oriented firms will have more complex and technical disclosures and greater incentives to obfuscate unfavourable information. Regulatory pronouncements to improve readability are less likely to achieve the desired outcome for such firms. While our study has focused on the entire 10-K readability, future studies can consider investigating the impact of business strategy on the readability of a specific section in the 10-K, for instance, MD&A where managers exhibit greater discretion in disclosures.

#### Data availability

All data used in this study are publicly available from the sources identified in the paper.

#### Acknowledgments

We would like to thank Larry Gordon and Martin Loeb and two anonymous reviewers for their constructive comments that assisted us to further develop and improve the paper.

#### References

Bagnoli, M., Watts, S., 2010. Oligopoly, disclosure, and earnings management. Account. Rev. 85, 1191-1214.

Barnett, A., Leoffler, K., 1979. Readability of accounting and auditing messages. J. Bus. Commun. 16, 49-59.

Barth, M.E., Kasznik, R., McNichols, M.F., 2001. Analyst coverage and intangible assets. J. Account. Res. 39, 1-34.

Bartlett, S., Chandler, R., 1997. The corporate report and the private shareholder: Lee and Tweedie twenty years on. Br. Account. Rev. 29, 245-261.

Beattie, V., 2014. Accounting narratives and the narrative turn in accounting research: issues, theory, methodology, methods and a research framework. Br. Account. Rev. 46, 111–134.

Beatty, A., Liao, S., Yu, J., 2013. The spillover effect of fraudulent financial reporting on peer firms' investments. J. Account. Econ. 55, 183–205.

Bentley, K.A., 2012. Antecedents to Financial Statement Misreporting: The Influence of Organizational Business Strategy, Ethical Culture and Climate. Doctoral dissertation, Texas A&M University, College Station.

Bentley, K.A., Omer, T.C., Sharp, N.Y., 2013. Business strategy, financial reporting irregularities, and audit effort. Contemp. Account. Res. 30, 780–817. Bentley-Goode, K.A., Newton, N.J., Thompson, A.M., 2017. Business strategy, internal control over financial reporting, and audit reporting quality. Aud. J. Practice Theory 36, 49–69.

Bentley-Goode, K.A., Omer, T.C., Twedt, D.J., 2018. Does business strategy impact a firm's information environment? J. Account. Audit. Financ forthcoming. Biddle, G., Hilary, G., Verdi, R., 2009. How does financial reporting quality relate to investment efficiency? J. Account. Econ. 48, 112–131. Bloomfield, R., 2002. The 'Incomplete Revelation Hypothesis' and financial reporting. Account. Horizons 16, 233–243. Bloomfield, R., 2008. Discussion of "Annual report readability, current earnings, and earnings persistence". J. Account. Econ. 45, 248–252.

#### EK Lim et al.

#### Journal of Accounting and Public Policy xxx (xxxx) xxx-xxx

Bonsall IV, S., Leone, A., Miller, B., Rennekamp, K., 2017. A Plain English measure of financial reporting readability. J. Account. Econ. 63, 329-357.

Burgelman, R.A., 2002. Strategy is Destiny: How strategy-making Shapes a Company's Future. The Free Press, New York.

Cameron, A., Miller, D., 2015. A practitioner's guide to cluster-robust inference. J. Hum. Resour. 50, 317-372.

- Carson, E., Fargher, N., 2007. Note on audit fee premiums to client size and industry specialization. Account. Financ. 47, 423-446.
- Cazier, R.A., Pfeiffer, R., 2016. Why are 10-K filings so long? Account Horizons 30, 1-21.

Chen, K.C.W., Chen, Z., Wei, K., 2011. Agency costs for free cash flows and the effect of shareholder rights on the implied cost of equity capital. J. Financ. Quant. Anal. 46, 171-207.

Chen, Y., Eshleman, J., Soileau, J., 2017. Business strategy and auditor reporting. Aud. J. Practice Theory 36, 63-86.

- Cheng, J.L.C., Kesner, I.F., 1997. Organizational slack and response to environmental shifts: the impact of resource allocation patterns. J. Manage. 23, 68-86.
- Child, J., 1972. Organizational structure, environment and performance: the role of strategic choice. Sociology 6, 2-22.

Clark, K.B., 1985. The interaction of design hierarchies and market concepts in technological evolution. Res. Policy 14 (5), 235-251.

Cohen, D., Dey, A., Lys, T., 2008. Real and accrual-based earnings management in the pre- and post-Sarbanes-Oxley periods. Account. Rev. 83, 757-787.

Courtis, J., 1982. Private shareholders' response to corporate annual reports. Account. Financ. 22, 53-72.

Courtis, J., 1998. Annual report readability variability: tests of the obfuscation hypothesis. Account. Audit. Account. J. 11, 459-471.

Courtis, J., 2004. Corporate report obfuscation: artefact or phenomenon? Br. Account. Rev. 36, 291-312.

Cox, C., 2007. Closing remarks to the Second Annual Corporate Governance Summit. < https://www.sec.gov/news/speech/2007/spch032307cc.htm > (accessed 16. 11.15).

Daines, R., 2001. Does Delaware law improve firm value? J. Financ. Econ. 62, 525-558.

Davis, A., Tama-Sweet, I., 2012. Managers' use of language across alternative disclosure outlets: earnings press release versus MD&A. Contem. Account. Res. 29, 804-837

Dechow, P., Sloan, R., Sweeney, A., 1995. Detecting earnings management. Account. Rev. 70, 193-225.

Dedman, E., Mouselli, S., Shen, Y., Stark, A., 2009. Accounting, intangible assets, stock market activity, and measurement and disclosure policy- Views from the U.K. Abacus 45, 312-341.

Dempsey, S.J., Harrison, D.M., Luchtenberg, K.F., Seiler, M.J., 2012. Financial opacity and firm performance: the readability of REIT annual reports. J. Real Estate Financ. Econ. 45, 450-470.

DeSarbo, W.S., Di Benedetto, A.C., Song, M., Sinha, I., 2005. Revisiting the Miles and Snow strategic framework: uncovering interrelationships between strategic types, capabilities, environmental uncertainty, and firm performance. Strateg. Manage. J. 26, 47-74.

Dess, G., Beard, D., 1984. Dimensions of organizational environments. Adm. Sci. Q. 29, 313-327.

Dhaliwal, D., Huang, S., Khurana, I., Pereira, R., 2014. Product market competition and conditional conservatism. Rev. Account. Stud. 19, 1309–1345.

Ecker, B., van Triest, S., Williams, C., 2013. Management control and the decentralization of R&D. J. Manage. 39, 906-927.

Ellis, J., Fee, C., Thomas, S., 2012. Proprietary costs and the disclosure of information about customers. J. Account. Res. 50, 685-727.

Ertugrul, M., Lei, J., Qiu, J., Wan, C., 2017. Annual report readability, tone ambiguity, and the cost of borrowing. J. Financ. Quant. Anal. 52, 811-836.

Fich, E.M., Shivdasani, A., 2006. Are busy boards effective monitors? J. Financ. 61, 689-724.

Financial Reporting Council, 2011. Cutting Clutter: Combating Clutter in Annual Reports, The Financial Reporting Council Limited, London.

Ghosh, D., Olsen, L., 2009. Environmental uncertainty and managers' use of discretionary accruals. Account. Organ. Soc. 34, 188-205.

Griffin, P.A., 2003. Got information? Investor response to Form 10-K and Form 10-O EDGAR filings. Rev. Account. Stud. 8, 433-460.

Grossman, S., Stiglitz, J., 1980. On the impossibility of informationally efficient markets. Am. Econ. Rev. 70 (3), 393-408.

Hambrick, D.C., 1983a. Some tests of the effectiveness and functional attributes of Miles and Snow's strategic types. Acad. Manage. J. 26, 5-26.

Hambrick, D.C., 1983b. High profit strategies in mature capital goods industries: a contingency approach. Acad. Manage. J. 26, 687-707.

Hambrick, D.C., MacMillan, I., Barbosa, R., 1983. Business unit strategy and changes in the product R&D budget. Manage. Sci. 29, 757-769.

Henry, E., Leone, A., 2016. Measuring qualitative information in capital markets research: comparison of alternative methodologies to measure disclosure tone. Account. Rev. 91, 153-178.

Higgins, D., Omer, T., Phillips, J., 2015. The influence of a firm's business strategy on its tax aggressiveness. Contem. Account. Res. 32, 674-702.

Hill, S., Martin, R., Harris, M., 2000. Decentralization, integration and the post-bureaucratic organization: the case of R&D. J. Manage. Stud. 37, 563-585.

Hirshleifer, D., Teoh, S., 2003. Limited attention, information disclosure, and financial reporting. J. Account. Econ. 36, 337-386.

Huang, C., Chu, P., Chiang, Y., 2008. A fuzzy AHP application in government-sponsored R&D project selection. Omega 36, 1038–1052.

Huddart, S., Ke, B., Shi, C., 2007. Jeopardy, non-public information, and insider trading around SEC 10-K and 10-Q filings. J. Account. Econ. 43, 3–36. Hynes, G., Bexley, J., 2004. Understandability of banks' annual reports. Educ. Rev. Bus. Commun. 1, 81-96.

Iannoconi, T., Sinnett, W., 2011. Disclosure overload and complexity: Hidden in plain sight. < https://www.kpmg.com/US/en/IssuesAndInsights/ ArticlesPublications/Documents/KPMG%20FEI%20Disclosure%20Report\_final.web.pdf > (accessed 16.11.15).

International Accounting Standards Board, 2017. Discussion Paper DP/2017/1 Disclosure Initiative – Principles of Disclosure, The IFRS Foundation, London. Ittner, C., Larcker, D., 1997. Quality strategy, strategic control systems and organizational performance. Account. Organ. Soc. 22, 293-314.

- Ittner, C., Larcker, D., 2001. Assessing empirical research in managerial accounting: a value-based management perspective. J. Account. Econ. 32, 349-410.

Ittner, C., Larcker, D., Rajan, M., 1997. The choice of performance measures in annual bonus contracts. Account. Rev. 72, 231-255.

Jaworski, B.J., Kohli, A.K., 1993. Market orientation: antecedents and consequences. J. Mark. 57, 53-70.

Jones, J.J., 1991. Earnings management during import relief investigations. J. Account. Res. 29 (2), 193-228.

Kennedy, P., 1992. A Guide to Econometrics. MIT Press, Cambridge.

Kintsch, W., Van Dijk, T.A., 1978. Toward a model of text comprehension and production. Psychol. Rev. 85, 363-394.

Klein, A., 1998, Firm performance and board committee structure, J. Law Econ. 41, 275-303.

Kohli, A.K., Jaworski, B.J., 1990. Market orientation: The construct, research propositions, and managerial implications. J. Mark. 54, 1–18.

Kotha, S., Nair, A., 1995. Strategy and environment as determinants of performance: evidence from the Japanese machine tool industry. Strateg. Manage. J. 16, 497–518.

Kothari, S.P., Li, X., Short, J., 2009. The effect of disclosures by management, analysts, and financial press on cost of capital, return volatility, and analyst forecasts: a study using content analysis. Account. Rev. 84, 1639-1670.

Kumar, G., 2014. Determinants of readability of financial reports of U.S. listed Asian companies. Asian J. Financ. Account. 6, 1-18.

Laksamana, I., Tietz, W., Yang, Y., 2012. Compensation discussion and analysis (CD&A) readability and management obfuscation. J. Account. Public Policy 31, 185-203.

Langerak, F., Nijssen, E., Frambach, R., Gupta, A., 1999. The influence of strategy on R&D knowledge domains and capabilities. R&D Manage. 29, 209-217.

Lawrence, A., 2013. Individual investors and financial disclosure. J. Account. Econ. 56, 130-147.

Lee, T., Tweedie, D., 1975. Accounting information: an investigation of private shareholder usage. Account. Bus. Res. 6, 280-291.

Lee, Y., 2012. The effect of quarterly report readability on information efficiency of stock prices. Contem. Account. Res. 29, 1137–1170.

Lehavy, R., Li, F., Merkley, K., 2011. The effect of annual report readability on analyst following and the properties of their earnings forecasts. Account. Rev. 86, 1087-1115.

Leifer, R., Mills, P., 1996. An information processing approach for deciding upon control strategies and reducing control loss in emerging organisations. J. Manage. 22, 113-137.

Levy, D.T., 1989. The transaction cost approach to vertical integration: an empirical examination. Rev. Econ. Stat. 67, 438-445.

Li, F., 2008. Annual report readability, current earnings, and earnings persistence. Fog data from Li (2008). http://webuser.bus.umich.edu/feng/. J. Account. Econ. 45, 221-247.

Li, F., 2010. Textual analysis of corporate disclosures. A survey of the literature. J. Account. Lit. 29, 143-164.

#### EK Lim et al

#### Journal of Accounting and Public Policy xxx (xxxx) xxx-xxx

Lo, K., Ramos, F., Rogo, 2017. Earnings management and annual report readability. J. Account. Econ. 63, 1-25.

- Loughran, T., McDonald, B., 2011. When is a liability not a liability? Textual analysis, dictionaries, and 10-Ks. J. Financ. 66, 35-65.
- Loughran, T., McDonald, B., 2013. IPO first-day returns, offer price revisions, volatility, and form S-1 language. J. Financ. Econ. 109, 307-326.

Loughran, T., McDonald, B., 2014. Measuring readability in financial disclosures. J. Financ. 69, 1643–1671.

Martikainen, M., Miihkinen, A., Watson, L., 2016. Board characteristics and disclosure tone. Working paper, Hanken School of Economics and University of Florida. Masson, M., Waldron, M., 1994. Comprehension of legal contracts by non-experts: effectiveness of plain language redrafting. Appl. Cogn. Psychol. 8, 67-85.

McDaniel, S., Kolari, J., 1987. Marketing strategy implications of the Miles and Snow strategic typology. J. Market. 51, 19-30.

Miles, R.E., Snow, C.C., 1978. Organizational Strategy, Structure, and Process. McGraw-Hill, San Francisco. Miles, R.E., Snow, C.C., 1984. Fit, failure and the hall of fame. Calif. Manage. Rev. 26, 10-28.

Miles, R.E., Snow, C.C., 1986. Organizations: New concepts for new forms. Calif. Manage. Rev. 28, 62-73.

Miles, R.E., Snow, C.C., 2003. Organizational Strategy, Structure, and Process. Stanford University Press, Stanford.

Miles, R.E., Snow, C.C., Meyer, A.D., Coleman, H.J., 1978. Organizational strategy, structure and process. Acad. Manage. Rev. 3, 546-562.

Milgrom, P., Roberts, J., 1995. Complementarities and fit: strategy, structure, and organizational change in manufacturing. J. Account. Econ. 19, 179-208.

Miller, B., 2010. The effects of reporting complexity on small and large investor trading. Account. Rev. 85, 2107-2143.

Miller, D., Ross, M., 1975. Self-serving biases in the attribution of causality: fact or fiction? Psychol. Bull. 82, 213-225.

Naiker, V., Navissi, F., Sridharan, V., 2008. The agency cost effects of unionization on firm value. J. Manage. Account. Res. 20, 133-152.

Navissi, F., Sridharan, V., Khedmati, M., Lim, E.K.Y., Evdokimov, E., 2017. Business strategy, over- (under-) investment and managerial compensation. J. Manage. Account. Res. 29, 63-86.

Pantzalis, C., Park, J., 2009. Equity market valuation of human capital and stock returns. J. Bank. Financ. 33, 1610–1623.

Park, K., 2017. Pay disparities within top management teams and earnings management. J. Account. Public Policy 36, 59-81.

Petersen, M.A., 2009. Estimating standard errors in finance panel data sets: comparing approaches. Rev. Financial Stud. 22 (1), 435-480.

Rennekamp, K., 2012. Processing fluency and investors' reactions to disclosure readability. J. Account. Res. 50, 1319-1354.

Rogers, J., Van Buskirk, A., Zechman, S., 2011. Disclosure tone and shareholder litigation. Account. Rev. 86, 2155-2183.

Roychowdhury, S., 2006. Earnings management through real activities manipulation. J. Account. Econ. 42 (3), 335–370.

Rutherford, B., 2003. Obfuscation, textual complexity and the role of regulated narrative accounting disclosure in corporate governance. J. Manage. Gov. 7, 187-210. Sadka, G., 2004. Financial reporting and product markets: Learning from competitors. Working paper, University of Chicago.

Said, A.A., HassabElnaby, H., Wier, B., 2003. An empirical investigation of the performance consequences of non-financial measures. J. Manag. Account. Res. 15, 193-223

Saleh, S.D., Wang, C.K., 1993. The management of innovation: Strategy, structure, and organizational climate. IEEE Trans. Eng. Manage. 40, 14–21.

Saviotti, P., 1998. On the dynamics of appropriability, of tacit and of codified knowledge. Res. Policy 26, 843-856.

Seavey, S., Imhof, M., Watanabe, O., 2017. Competition, proprietary costs of financial reporting and financial statement competition. Working paper, University of Nebraska-Lincoln, Wichita State University, and Iowa State University.

Securities and Exchange Commission (SEC), 1998. A Plain English Handbook: How to Create Clear SEC Disclosure Documents, SEC, Washington.

Segev, E., 1989. A systematic comparative analysis of two business-level strategic typologies. Strateg. Manage. J. 10, 487-505.

Shah, A.K., Oppenheimer, D.M., 2007. Easy does it: the role of fluency in cue weighting. Judgm. Decis. Mak. 2 (6), 371-379.

Shortell, S., Zajac, E., 1990. Perceptual and archival measures of Miles and Snow's strategic types: a comprehensive assessment of reliability and validity. Acad. Manage, J. 33, 817-832.

Smith, K., Guthrie, J., Chen, J., 1989. Strategy, size and performance. Organ. Stud. 10, 63-81.

Snow, C., Hambrick, D., 1980. Measuring organizational strategies: Some theoretical and methodological problems. Acad. Manage. Rev. 5, 527-538.

Song, M., Montoya-Weiss, M.M., 2001. The effects of perceived technological uncertainty on Japanese new product development. Acad. Manage. J. 44, 61-80. Su, Z., Xie, E., Peng, J., 2010. Impacts of environmental uncertainty and firm's capabilities on R&D investment: evidence from China. Innov. Manage. Policy Pract. 12,

269-282. Tansey, P., Spillane, J., Meng, X., 2014. Linking response strategies adopted by construction firms during the 2007 economic recession to Porter's generic strategies.

Constr. Manage. Econ. 32, 705-724. Thomas, A., Ramaswamy, K., 1996. Matching managers to strategy: further tests of the Miles and Snow typology. Br. J. Manage. 7, 247-261.

Wagenhofer, A., 1990. Voluntary disclosure with a strategic opponent. J. Account. Econ. 12, 341-363.

White, M.J., 2013. The path forward on disclosure. < http://www.sec.gov/News/Speech/Detail/Speech/1370539878806#.Uzx2VPldV8E > (accessed 16.11.15). Williamson, O., 1979. Transaction cost economics: the governance of contractual relations. J. Law Econ. 22, 233-261.

You, H., Zhang, X., 2009. Financial reporting complexity and investor under-reaction to 10-K information. Rev. Account. Stud. 14, 559-586.

Zahra, S.A., Priem, R.L., Rasheed, A.A., 2005. The antecedents and consequences of top management fraud. J. Manage. 31, 803-828.

Zajac, E., Shortell, S., 1989. Changing generic strategies: Likelihood, direction and performance implications. Strateg. Manage. J. 10, 413–430.